

*Citation for published version:* Schiffer, M, Schneider, M, Walther, G & Laporte, G 2019, 'Vehicle routing and location routing with intermediate stops: A review', Transportation Science, vol. 53, no. 2, pp. 319-343. https://doi.org/10.1287/trsc.2018.0836

DOI: 10.1287/trsc.2018.0836

Publication date: 2019

Document Version Peer reviewed version

Link to publication

Schiffer, M., Schneider, M., Walther, G., & Laporte, G. (2019). Vehicle routing and location routing with intermediate stops: A review. Transportation Science, 53(2), 319-343. https://doi.org/10.1287/trsc.2018.0836

**University of Bath** 

## **Alternative formats**

If you require this document in an alternative format, please contact: openaccess@bath.ac.uk

#### **General rights**

Copyright and moral rights for the publications made accessible in the public portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognise and abide by the legal requirements associated with these rights.

Take down policy If you believe that this document breaches copyright please contact us providing details, and we will remove access to the work immediately and investigate your claim.

### Submitted to Transportation Science manuscript TS-2017-0178

Authors are encouraged to submit new papers to INFORMS journals by means of a style file template, which includes the journal title. However, use of a template does not certify that the paper has been accepted for publication in the named journal. INFORMS journal templates are for the exclusive purpose of submitting to an INFORMS journal and should not be used to distribute the papers in print or online or to submit the papers to another publication.

# Vehicle routing and location-routing with intermediate stops: A review

#### Maximilian Schiffer

Chair of Operations Management, School of Business and Economics, RWTH Aachen University Kackertstraße 7, 52072 Aachen, Germany, maximilian.schiffer@om.rwth-aachen.de, www.om.rwth-aachen.de

#### Michael Schneider

Deutsche Post Chair of Optimization of Distribution Networks, School of Business and Economics, RWTH Aachen University Kackertstraße 7, 52072 Aachen, Germany, schneider@dpo.rwth-aachen.de, www.dpo.rwth-aachen.de

#### Grit Walther

Chair of Operations Management, School of Business and Economics, RWTH Aachen University Kackertstraße 7, 52072 Aachen, Germany, walther@om.rwth-aachen.de, www.om.rwth-aachen.de

#### Gilbert Laporte

CIRRELT and Canada Research Chair in Distribution Management, HEC Montréal, 3000, chemin de la Côte-Sainte-Catherine, Montréal, Canada H3T 2A7, gilbert.laporte@cirrelt.ca, http://chairedistributique.hec.ca/en/

This paper reviews the literature on vehicle routing problems and location-routing problems with intermediate stops. We classify publications into different categories from both an application-based perspective and a methodological perspective. In addition, we analyze the papers with respect to the algorithms and benchmark instances they present. Furthermore, we provide an overview of trends in literature and identify promising areas for further research.

*Key words*: intermediate stops, intra-route facilities, vehicle routing, location-routing, survey *History*:

### 1. Introduction

Routing problems with intermediate stops (RPIS) are highly relevant in logistics, arising, e.g., in freight transportation and municipal waste collection. However, the interest of the research community in these problems has for a long time mainly been limited to single application cases. In recent years, researchers have started to pay increasing attention to RPIS, especially regarding applications in city logistics and logistics fleets with alternative fuel vehicles (AFVs). The purpose of this review is to guide the reader through the literature on RPIS by providing a discussion of research fields, application cases, and methodological classifications. We assume that the reader has a basic knowledge of vehicle routing (cf. Golden, Raghavan, and Wasil 2008, Toth and Vigo 2014), location theory (cf. Daskin 1995, Laporte et al. 2015), and exact and heuristic solution methods for combinatorial optimization problems in general (cf. Wolsey 1998, Desaulniers, Desrosiers, and Solomon 2005, Gendreau and Potvin 2010).

We start the discussion with a precise definition of the terms *intermediate stop* and *intra-route facility*. These definitions are necessary to characterize the considered problem class, but they are ambiguously used in the literature and in different research streams. In this work, an intermediate stop is an optional stop en route in order to keep a vehicle operational while fulfilling its main service task. Thus, an intermediate stop differs from a regular stop (e.g., providing service at a customer) and also from an optional customer stop that, e.g., arises in vehicle routing problems (VRPs) with profits (cf. Archetti, Speranza, and Vigo 2014). Although intermediate stops are optional, they may prove unavoidable to keep vehicles operational. Intermediate stops take place at so-called intra-route facilities, which enable vehicles to replenish a certain resource. An intra-route facility is planned at the same echelon as the customers. It is visited en route and thus differs from a so-called intermediate facility, which is often used as a synonym for a depot or a hub in multi-echelon and cross-docking operations. Therefore, we exclude work on multi-echelon routing problems from this survey and refer to Guastaroba, Speranza, and Vigo (2016) for a deep overview of the topic.

Before detailing the aim and organization of this survey, we first outline application areas in which RPIS arise. Next, we show how these problems can be categorized from a modeling point of view cutting across the described application areas. Both, the application-based and the problembased classification are then used to organize this survey in a concise fashion.

#### 1.1. Application areas

Since the early 1970s, researchers have studied RPIS. While this research was very sparse until the year 2000, it increased significantly from then on due to new challenges arising in city logistics and AFV fleets. More general, RPIS arise in three main application areas:

i) Replenishment and disposal of goods or waste: In certain distribution networks, satellite facilities are used to avoid deadheads caused by return trips to the depot to replenish freight (cf. Angelelli and Speranza 2002b, Crevier, Cordeau, and Laporte 2007, Tarantilis, Zachariadis, and Kiranoudis 2008). Real-world examples for such a distribution structure can be found in heating oil distribution (Prescott-Gagnon, Desaulniers, and Rousseau 2014), road maintenance (Amaya, Langevin, and Trépanier 2007), and city logistics (Crainic, Ricciardi, and Storchi 2009). Analogous concepts can be used to dispose of freight in collection problems. Two main application areas belong into this context, namely waste collection (cf. Kim, Kim, and Sahoo 2006, Benjamin and Beasley 2010) and snow plowing (cf. Perrier, Langevin, and Campbell 2006, Salazar-Aguilar, Langevin, and Laporte 2012).

- ii) Refueling: Routing problems with refueling stops are encountered in dense or in sparse refueling network structures. In dense network structures, problems arise for economic reasons (e.g., company contracts with lower prices or large price differences between stations located in close vicinity). If AFVs are used as sustainable means of transportation, the necessary refueling infrastructure for such new technologies is often still sparse. In addition, the driving range of several types of AFVs, e.g., electric commercial vehicles (ECVs), is limited. Thus, refueling stops have to be considered explicitly in the respective routing problems.
- iii) Idling for rest periods and breaks: Focusing on long-haul distribution or multi-day trips, intermediate stops for idling times take place because of hours of service (HOS) regulations in freight transportation (Goel 2009) or due to hotel selection (Vansteenwegen, Souffriau, and Sörensen 2012) to prevent drivers' fatigue.

The research interest in these three application areas evolved differently in the last years. Figure 1 shows this development by illustrating the total number of publications over time. Intermediate stops for replenishing or unloading were studied first, and the number of papers on this topic is constantly rising. The papers mainly focus on conventional applications in municipal services (e.g., waste collection) and logistics systems (e.g., freight replenishment), which have been and still are relevant for practitioners. Up to 2010, papers on refueling issues were mainly limited to routing of internal combustion engine vehicles (ICEVs) in a dense infrastructure. Due to a significant interest of researchers and practitioners in routing problems with AFVs, especially ECVs, we can observe a huge increase in the number of papers from then on. Publications on intermediate stops for idling



Figure 1 Total number of publications over time for the three application areas.

due to HOS regulations and hotel selection in routing problems experienced the strongest increase between the year 2000 and 2015.

#### 1.2. Problem categorization

RPIS can also be classified according to the characteristics of different problem variants (cf. Schiffer et al. 2017b). Common characteristics are time constraints, i.e., problem variants with a maximum route duration and problem variants restricted by time windows at customers. RPIS can be further differentiated with respect to the characteristics of the replenishment process, which is closely linked to overall time constraints if replenishment is time-consuming. In addition, the replenished resource (from here on referred to as the operational resource) can differ with respect to its consumption. Detailing the characteristics listed above, the following nomenclature to characterize RPIS can be derived:

- *i)* Time constraints can be given either by time windows (TW), by a maximum route duration (MRD), or can be neglected (none).
- *ii*) *Replenishment time* can either be dependent (D) or independent (I) of the quantity of the operational resource that has to be replenished.
- *iii*) *Replenishment processes* can either be restricted to full replenishment (F) or may allow partial replenishment (P).
- *iv)* Operational resources can be characterized with regard to the type of consumption of the operational resource, which can either be node-based (N) (e.g., freight) or arc-based (A) (e.g., fuel).

The resulting nomenclature is shown in Table 1. Problems are separated into node-based and arc-based problems in the first dimension. Further, problems can be characterized by the type of time restrictions. The second dimension divides these problems further into quantity-dependent and quantity-independent replenishment processes. Quantity-dependent processes can be further separated into full and partial replenishment processes. Partial replenishment is not considered for time-independent models because the time savings it yields are irrelevant.

	т	able 1 Nomencla	ture of RPIS variants.			
		Dependent	replenishment	Independent replenishmen		
Consumption	Time restriction	Full replenishment	Partial replenishment	Full replenishment		
	none	NDF	NDP	NIF		
node-based	route duration	NDFMRD	NDPMRD	NIFMRD		
	time windows	NDFTW	NDPTW	NIFTW		
	none	ADF	ADP	AIF		
arc-based	route duration	ADFMRD	ADPMRD	AIFMRD		
	time windows	ADFTW	ADPTW	AIFTW		

In analogy to Schiffer et al. (2017b).

#### 1.3. Aim and organization

The contribution of this survey is twofold. First, the past, current and future application areas of RPIS are highlighted to provide researchers with a knowledge of the development of this research field over time. Second, methodological enhancements and problem-specific details are discussed. This survey contains a dedicated section for each application area. Within these sections, the application areas are further separated into sections on specific research streams and one analysis section for each application area that summarizes its main findings. The nomenclature presented in Table 1 is used in the analysis section to highlight the focus of each application area and research stream. We pay special attention to the discussion of the following aspects: i) We analyze the scope of each publication, mainly dividing between case studies and algorithmic contributions. In this context, we categorize a publication as a case study paper if it comprises a real-world dataset and if the related problem is of interest for practitioners. Thus, papers that derive only instance sets from real-world data without any context or discussion are regarded as methodological papers. Furthermore, we subdivide publications into arc routing and node routing problems, and we discuss the type of data with respect to deterministic and uncertain information. ii) We analyze the various objectives used within the application field and discuss their relevance. *iii*) To show which algorithms are most suitable and popular for the specific application area, we discuss the different solution methods that have been developed. This discussion is split into a part on exact algorithms and a part on metaheuristics. iv) We provide information on the available benchmarks for each research steam and provide a collection of publicly available instances at www.om.rwth-aachen.de/ data/litrevInst. Finally, we summarize the main findings for each application area and discuss future research questions.

The remainder of this paper is structured as follows. Section 2 reviews literature on intermediate stops for replenishment and unloading. Section 3 focuses on intermediate stops for refueling. Section 4 analyzes literature on intermediate stops for idling due to breaks or rests. Section 5 concludes this survey by summarizing its main findings. In the appendix, we provide a glossary of the abbreviations used in this paper.

#### 2. Intermediate stops for replenishment or unloading of goods

In this section, we focus on publications that address intermediate stops for the replenishment or the unloading of goods. Classical application cases often arise in retail and distribution logistics and municipal services. Besides these classical problems, selected VRPs with synchronization constraints represent RPIS according to our definition in Section 1. Thus, Section 2.1 focuses on intermediate stops for replenishment, Section 2.2 focuses on intermediate stops for unloading, and synchronization problems are discussed in Section 2.3. In Section 2.4, we conduct an analysis of problem characteristics and solution methods.

#### 2.1. Intermediate stops for goods replenishment

Intermediate stops for goods replenishment often arise in distribution networks for raw materials or in small package shipping.

The first publication that introduced intermediate stops in this context focused on propane gas distribution. Bard et al. (1998) addressed the VRP with satellite facilities and developed a branchand-cut (BC) algorithm for this problem. The objective was to minimize the overall distance under a maximum tour duration constraint. Another application in raw material distribution was discussed by Prescott-Gagnon, Desaulniers, and Rousseau (2014) who proposed a VRP arising in heating oil distribution, considering intra-route replenishments, heterogeneous vehicles, optional customer visits and time windows. The authors designed a tabu search (TS) heuristic, a large neighborhood search (LNS) heuristic with a TS component, and a column generation (CG) metaheuristic to analyze a real-world instance.

In parcel logistics, several publications addressed arc routing and node routing problems. Focusing on arc routing problems, Ghiani and co-authors investigated the capacitated arc routing problem (CARP) with intermediate facilities (CARPIF) in several publications. Ghiani, Improta, and Laporte (2001) introduced the CARPIF as an extension of the pure CARP, accounting for intermediate stops for replenishment or unloading. The authors presented a lower bound based on the rural postman problem and a linear integer program. Ghiani et al. (2004) extended the CARPIF with capacity and length restrictions and developed three heuristics, namely a construction algorithm and two TS algorithms. Ghiani et al. (2010) provided an ant colony algorithm for the CARPIF that outperformed existing algorithms. Polacek et al. (2008) presented a variable neighborhood search (VNS) for the CARP with refill points (CARPRP) that was capable of finding the best known solution (BKS) for all 120 instances of four different benchmark sets for the CARP and the CARPRP, and improved 71 BKSs.

Focusing on node routing problems, Angelelli and Speranza (2002b) extended the periodic VRP (PVRP) to intermediate facilities. Minimizing the overall traveled distance, they proposed a TS heuristic for this problem and presented results on instances with 50–288 customers. Crevier, Cordeau, and Laporte (2007) introduced the multi-depot VRP with inter-depot routes (MDVRPI), which considers intermediate depots at which vehicles can be replenished during the course of a route. Tarantilis, Zachariadis, and Kiranoudis (2008) renamed this problem to the VRP with intermediate replenishment facilities (VRPIRF) and proposed a hybrid guided local search (LS) heuristic. Kek, Cheu, and Meng (2008) studied a capacitated VRP (CVRP) with flexible start and end depots, allowing for intermediate replenishment visits to any depot. The authors presented a mixed integer program to minimize travel and vehicle costs and found that cost savings of 49% can be reached for a specific case study in Singapore. Muter, Cordeau, and Laporte (2014) developed

a branch-and-price algorithm for the MDVRPI. The authors discussed the benefit of two different pricing subproblems and managed to solve problem instances with up to 50 customers to optimality.

Recent publications focused on generic algorithmic frameworks for RPIS. Schneider, Stenger, and Hof (2015) introduced the VRP with intermediate stops (VRPIS) and provided results on different problems. Schiffer et al. (2017b) focused on RPIS by analyzing different types of resources and replenishment options, and presented an algorithmic framework that yields new BKSs for most existing problem variants. Both publications also investigated the MDVRPI benchmarks of Crevier, Cordeau, and Laporte (2007) and Tarantilis, Zachariadis, and Kiranoudis (2008).

#### 2.2. Intermediate stops for unloading of goods

Intermediate stops for unloading of goods arise in municipal service applications, especially in waste collection problems. For an in-depth overview on waste collection problems that does not only cover problems related to intermediate stops, we refer to Beliën, De Boeck, and Van Ackere (2014).

The first publication on intermediate stops for unloading is due to Beltrami and Bodin (1974) focusing on the routing of waste collection vehicles with disposal facilities for a real-world problem arising in New York and Washington. Mourão and Almeida (2000) investigated a CARP with intermediate stops for a household refuse problem in Lisbon. A lower bound and a route-first cluster-second heuristic were presented, and the algorithm was tested on a benchmark set based on the real-world case. Mourão and Amado (2005) presented another heuristic for this problem based on a multi-graph representation, which improved their previous results and performed well on largesized instances with up to 400 nodes and 1215 arcs. Angelelli and Speranza (2002a) applied the periodic VRP with intermediate facilities (PVRP) originally presented in Angelelli and Speranza (2002b) to case studies arising in waste collection. The authors presented a TS heuristic to solve large instances. De Rosa et al. (2002) introduced the arc routing and scheduling problem with transshipment as a variant of the CARPRP. The problem arises in urban waste collection, where a fleet of vehicles recollects garbage, which is delivered to transfer stations, processed into compact units, and then transported to its final destination by trucks. Ghiani et al. (2005) applied the CARPIF to a waste collection problem in southern Italy, presenting a cluster-first route-second heuristic. Del Pia and Filippi (2006) studied a real-world case on waste collection in northern Italy using a CARP with intermediate stops. The authors implemented a VNS algorithm and found that a significant reduction in overall time (approx. 30%) can be achieved compared to the current realworld solution. Another real-world case of a waste management company was studied by Kim, Kim, and Sahoo (2006), who extended Solomon's insertion algorithm to this problem. Besides the case study, an instance set for the VRP with time windows (VRPTW) was considered. Santos, Coutinho-Rodrigues, and Current (2008) investigated a CARP with intermediate stops at drop-off points for a waste collection problem in Portugal. The authors implemented a decision support system based on a path-scanning algorithm. Benjamin and Beasley (2010) focused on the waste collection VRP with multiple disposal facilities and considered time windows and driver rest periods. Coene, Arnout, and Spieksma (2010) discussed a PVRP in the context of waste collection and presented a route-first cluster-second algorithm. Buhrkal, Larsen, and Ropke (2012) focused on a waste collection VRP in a city logistics context, considering time windows and minimizing the overall costs. In addition to numerical studies on existing benchmarks, the authors provided a case study of a Danish garbage company and proved that their algorithm is capable of improving the realworld results. Hemmelmavr et al. (2013a) and Hemmelmavr et al. (2013b) also studied the PVRP in the context of waste collection. The authors introduced a hybrid solution approach consisting of a VNS with a dynamic programming (DP) component. This solution procedure outperformed the approaches of Crevier, Cordeau, and Laporte (2007) and Tarantilis, Zachariadis, and Kiranoudis (2008) on the MDVRPI instances. Markov, Varone, and Bierlaire (2016) studied the waste collection VRP with intermediate facilities and investigated the impact of a heterogeneous fleet and flexible destination depots. A case study based on data of a waste company in Switzerland was presented, and the developed VNS obtains a mean improvement of 14.46% on the real-world solution. Willemse and Joubert (2016) investigated four different construction heuristics for the mixed CARPRP under time restrictions, aiming to identify a suitable heuristic for real-time support in real-world application cases.

Single publications on intermediate stops for unloading focused on other topics than waste collection. Jordan (1987) investigated a VRP with additional backhauls that can be stored at intermediate facilities instead of the home-depot. The authors presented a matching problem and a greedy heuristic to solve this problem. Perrier, Langevin, and Campbell (2007) discussed VRPs with intermediate stops for unloading operations in snow plowing.

#### 2.3. Intermediate stops for synchronization

Routing problems with synchronization constraints cover a wide range of application fields. Therefore, we limit the following discussion to synchronization problems that include an intermediate stop according to our definition in Section 1, and we refer to Drexl (2012) for an extensive overview on synchronization problems in general. More precisely, we restrict ourselves to problems in which the synchronization is limited to locations at a single echelon and occurs en route. Thus, multi-echelon synchronization problems (e.g., Contardo, Hemmelmayr, and Crainic 2012) are not considered. According to our definition in Section 1, intermediate stops that are directly related to providing service are not considered. This means that we also exclude dial-a-ride problems (e.g., Gørtz, Nagarajan, and Ravi 2009) and school bus routing (e.g., Fügenschuh 2009) from our analysis. Furthermore, we exclude publications on staff and driver scheduling (e.g., Dohn, Kolind, and Clausen 2009) because here intermediate stops are neither linked to an intra-route facility nor to a support vehicle, which can be seen as mobile intra-route facilities.

The publications discussed in the following can be separated into truck and trailer routing problems (TTRPs) and other routing problems with synchronization constraints. TTRPs are routing problems in which some customers can be served by trucks carrying a trailer, and other customers can only be served by a truck without a trailer. Thus, trailers can be parked and later be picked up by trucks if needed and allowed. In TTRPs, the parking space for trailers can be seen as an intra-route facility.

An overview on TTRPs can be found in Drexl (2013) and in Cuda, Guastaroba, and Speranza (2015). The first publication focusing on trailers in a VRP is due to Semet and Taillard (1993). The authors analyzed a real-world application on a grocery store distribution network and presented a TS heuristic. Gerdessen (1996) discussed the VRP with trailers allowing trucks to leave the trailer at a parking space and developed construction heuristics as well as a LS to solve the problem. Chao (2002) presented a TS heuristic for the TTRP. The algorithm was evaluated on instances with up to 150 customers. Scheuerer (2006) presented two construction heuristics and a TS heuristic for the TTRP. The algorithm outperformed the results of Chao (2002) on all instances. Tan, Chew, and Lee (2006) focused on the TTRP, investigating a multi-objective function, minimizing the distance and the number of trucks using an evolutionary algorithm. Villegas et al. (2010) presented a greedy randomized adaptive search procedure (GRASP), and a VNS with evolutionary LS for the single TTRP with time windows. Caramia and Guerriero (2010) presented a matheuristic for the TTRP based on a mixed integer program (MIP) and a LS procedure. Villegas et al. (2011) provided a combination of GRASP, VNS, and path relinking for the TTRP, which outperformed all previous algorithms. Derigs, Pullmann, and Vogel (2013) focused on the TTRP and discussed the impact of time windows and load transfers between trucks and trailers. Villegas et al. (2013) presented a matheuristic for the TTRP, consisting of a GRASP with iterated LS (ILS) and a set partitioning formulation. Drexl (2014) presented five different BC algorithms for the TTRP with transshipments and evaluated them on a large set of benchmark instances derived from real-world problems. Belenguer et al. (2016) discussed the single TTRP and included satellite facilities at which the trailer must be parked. The authors presented a BC algorithm capable of solving instances with up to 100 customers and 20 satellite facilities. Rothenbächer, Drexl, and Irnich (2016) developed a branch-and-price-and-cut (BPC) algorithm for the TTRP with time windows, considering quantitydependent transfer time. It outperformed existing approaches on known benchmark instances and was also applied to two real-world problems. Parragh and Cordeau (2017) focused on the TTRP with time windows in the context of infrastructure service providers that operate in urban areas. The authors developed a branch-and-price (BP) algorithm and an adaptive LNS (ALNS) to create initial columns and they manage to solve instances with up to 100 customers to optimality. Bartolini and Schneider (2017) develop a branch-and-cut algorithm based on a two-commodity flow formulation for the TTRP.

In other synchronization problems, the interchange of freight is conducted between two vehicles directly. In that case, designated support vehicles are used as mobile intra-route facilities. Such problems arise in municipal as well as logistics services. Amaya, Langevin, and Trépanier (2007) introduced the CARPRP in the context of road painting. Here, a vehicle that provides service on arcs is refilled at certain service points (in this case road junctions) by a second vehicle. In Amaya, Langevin, and Trépanier (2010), the authors presented a route-first cluster-second heuristic and a cutting-plane algorithm for the CARPRP and extended it to multiple loads. Using this heuristic, they solved a real-world case arising for road painting in Quebec. Salazar-Aguilar, Langevin, and Laporte (2012) studied an arc routing problem with synchronization constraints for snow plowing vehicles and presented an ALNS to study real-world large-sized instances. Salazar-Aguilar, Langevin, and minimize the makespan. The authors developed an ALNS that provided good results on a large set of artificial instances.

Note that we do not consider pickup and delivery problem with transshipments (see, e.g., Rais, Alvelos, and Carvalho 2014) as RPIS because the main purpose of transferring load at dedicated transshipment locations in these problems is to save travel costs and not to keep the vehicles operational by freeing capacity.

#### 2.4. Analysis of intermediate stops for replenishment and unloading

We now outline the characteristics of the publications discussed in Sections 2.1–2.3 with respect to their overall scope, objectives, and algorithmic contributions. Table 2 shows the scope of these papers, differentiating them according to i) the type of contribution (case study vs. methodological), ii) the type of the routing problem (node routing vs. arc routing), iii) the type of replenishment (unloading vs. replenishment vs. synchronization), and iv) the type of data (deterministic vs. uncertain).

As can be seen from Figure 2, a large majority of the publications focus on a methodological contribution (59%), while only 41% focus on application cases. Detailing this ratio in Table 2, it can be seen that most case studies are presented in the context of municipal services (e.g., waste collection, snow plowing), while publications addressing goods replenishment for classical logistics services often focus on algorithmic enhancements. The ratio between node routing (68%) and arc routing (32%) also indicates the share of different application cases in the analyzed publications. Arc routing problems are mainly discussed for municipal operations, and thus are related to

Table 2

Case Study Be Methodological Jo	eltrami and Bodin (1974), Semet and Taillard (1993), Mourão and Almeida (2000), Angelelli and Speranza (2002a), Ghiani et al. (2005), Mourão and Amado (2005), Del Pia and Filippi (2006), Kim, Kim, and Sahoo (2006), Kek, Cheu, and Meng (2008), Santos, Coutinho-Rodrigues, and Current (2008), Amaya, Langevin, and Trépanier (2010), Coene, Arnout, and Spieksma (2010), Buhrkal, Larsen, and Ropke (2012), Salazar-Aguilar, Langevin, and Laporte (2012), Hemmelmayr et al. (2013b), Prescott-Gagnon, Desaulniers, and Rousseau (2014), Markov, Varone, and Bierlaire (2016), Willemse and Joubert (2016) ordan (1987), Gerdessen (1996), Bard et al. (1998), Ghiani, Improta, and Laporte (2001), Angelelli and Speranza (2002b), Chao (2002), De Rosa et al. (2002), Ghiani et al. (2004), Scheuerer (2006), Tan, Chew, and Lee (2006), Amaya, Langevin, and Trépanier (2007), Crevier, Cordeau, and Laporte (2007), Polacek et al. (2008), Tarantlis, Zachariadis, and Kiranoudis (2008), Benjamin and Beasley (2010), Ghiani et al. (2010), Villegas et al. (2010, 2011), Derig, Pullmann, and Vogel (2013), Hemmelmayr et al. (2013a), Salazar-Aguilar, Langevin, and Laporte (2013), Drexl (2014), Muter, Cordeau, and Laporte (2014), Schneider, Stenger, and Hof (2015), Belenguer et al. (2016), Rothenbächer, Drexl, and Irnich (2016), Willemse and Joubert (2016), Bartolini and Schneider (2017), Parragh and Cordeau (2017)
Node routing Be	eltrami and Bodin (1974), Jordan (1987), Semet and Taillard (1993), Gerdessen (1996), Bard et al. (1998), Angelelli and Speranza (2002a,b), Chao (2002), Kim, Kim, and Sahoo (2006), Scheuerer (2006), Tan, Chew, and Lee (2006), Crevier, Cordeau, and Laporte (2007), Kek, Cheu, and Meng (2008), Tarantilis, Zachariadis, and Kiranoudis (2008), Benjamin and Beasley (2010), Coene, Arnout, and Spieksma (2010), Villegas et al. (2010, 2011), Buhrkal, Larsen, and Ropke (2012), Derigs, Pullmann, and Vogel (2013), Hemmelmayr et al. (2013a,b), Salazar-Aguilar, Langevin, and Laporte (2013), Drexl (2014), Muter, Cordeau, and Laporte (2014), Prescott-Gagnon, Desaulniers, and Rousseau (2014), Schneider, Stenger, and Hof (2015), Belenguer et al. (2016), Markov, Varone, and Bierlaire (2016), Rothenbächer, Drexl, and Irnich (2016), Bartolini and Schneider (2017), Parragh and Cordeau (2017) Iourão and Almeida (2000), Ghiani, Improta, and Laporte (2001), De Rosa et al. (2002), Ghiani et al. (2004, 2005), Mourão and Amado (2005), Del Pia and Filippi (2006), Amaya, Langevin, and Trépanier (2007), Polacek et al. (2008), Santos, Coutinho-Rodrigues, and Current (2008), Amaya, Langevin, and Trépanier (2010), Ghiani et al. (2010), Salazar-Aguilar, Langevin, and Laporte (2012, 2013), Willemse and Joubert (2016)
Replenishment Ba	ard et al. (1998), Ghiani, Improta, and Laporte (2001), Ghiani et al. (2004), Tan, Chew, and Lee (2006), Crevier, Cordeau, and Laporte (2007), Kek, Cheu, and Meng (2008), Tarantilis, Zachariadis, and Kiranoudis (2008), Ghiani et al. (2010), Muter, Cordeau, and Laporte (2014), Prescott-Gagnon, Desaulniers, and Rousseau (2014), Schneider, Stenger, and Hof (2015)
Unloading Be	eltrami and Bodin (1974), Jordan (1987), Mourão and Almeida (2000), Ghiani, Improta, and Laporte (2001), Angelelli and Speranza (2002b,a), De Rosa et al. (2002), Ghiani et al. (2004, 2005), Mourão and Amado (2005), Del Pia and Filippi (2006), Kim, and Sahoo (2006), Polacek et al. (2008), Santos, Coutinho-Rodrigues, and Current (2008), Benjamin and Beasley (2010), Coene, Arnout, and Spieksma (2010), Ghiani et al. (2010), Buhrkal, Larsen, and Ropke (2012), Hemmelmayr et al. (2013b,a), Markov, Varone, and Bierlaire (2016), Willemse and Joubert (2016).
Synchronization Se	Senet and Taillard (1993), Gerdessen (1996), Chao (2002), Scheuerer (2006), Amaya, Langevin, and Trépanier (2007), Amaya, Langevin, and Trépanier (2010), Caramia and Guerriero (2010), Villegas et al. (2010, 2011), Drexl (2012), Salazar-Aguilar, Langevin, and Laporte (2012), Derigs, Pullmann, and Vogel (2013), Drexl (2013), Salazar-Aguilar, Langevin, and Laporte (2013), Villegas et al. (2013), Drexl (2014), Belenguer et al. (2016), Rothenbächer, Drexl, and Irnich (2016), Bartolini and Schneider (2017), Parragh and Cordeau (2017)
Deterministic Be	eltrami and Bodin (1974), Jordan (1987), Semet and Taillard (1993), Gerdessen (1996), Bard et al. (1998), Mourão and Almeida (2000), Ghiani, Improta, and Laporte (2001), Angelelli and Speranza (2002b,a), Chao (2002), De Rosa et al. (2002), Ghiani et al. (2004, 2005), Mourão and Amado (2005), Del Pia and Filippi (2006), Kim, Kim, and Sahoo (2006), Scheuerer (2006), Tan, Chew, and Lee (2006), Amaya, Langevin, and Trépanier (2007), Crevier, Cordeau, and Laporte (2007), Kek, Cheu, and Meng (2008), Polacek et al. (2008), Santos, Coutinho-Rodrigues, and Current (2008), Tarantilis, Zachariadis, and Kiranoudis (2008), Amaya, Langevin, and Trépanier (2010), Benjamin and Beasley (2010), Coene, Arnout, and Spieksma (2010), Ghiani et al. (2010), Villegas et al. (2010, 2011), Buhrkal, Larsen, and Ropke (2012), Salazar-Aguilar, Langevin, and Laporte (2013), Derigs, Pullmann, and Vogel (2013), Hemmelmayr et al. (2013a,b), Salazar-Aguilar, Langevin, and Laporte (2013), Drexl (2014), Muter, Cordeau, and Laporte (2014), Prescott-Gagnon, Desaulniers, and Rousseau (2014), Schneider, Stenger, and Hof (2015), Belenguer et al. (2016), Markov, Varone, and Bierlaire (2016), Rothenbächer, Drexl, and Irnich (2016), Willemse and Joubert (2016), Bartolini and Schneider (2017), Parragh and Cordeau (2017)
Stochastic No	000

Scope of publications on intermediate stops for replenishment and unloading of goods.



Figure 2 Characteristics of publications on intermediate stops for replenishment and unloading.

road services or maintenance, e.g., snow plowing (Perrier, Langevin, and Campbell 2007), waste collection (Del Pia and Filippi 2006), and road painting (Amaya, Langevin, and Trépanier 2007).

Node routing problems are mainly discussed for classical logistics applications in which the service operation is related to single customer locations. Most publications focus on unloading (40%) or synchronization (40%), while only 21% focus on replenishment operations (cf. Figure 2). However, some of the synchronization problems arise in replenishment operations, e.g., (Amaya, Langevin, and Trépanier 2007, Amaya, Langevin, and Trépanier 2010). While a large majority of the publications focusing on unloading is related to waste collection, additional application cases for unloading problems arise in pickup problems in logistics networks, e.g., in milk collection (cf. Rothenbächer, Drexl, and Irnich 2016). Focusing on the type of data, none of the publications addresses uncertain data.

To describe the characteristics of the proposed problems with respect to time and replenishment or unloading restrictions, Figure 3 and Table 3 categorize the problem variants as outlined in Section 1. Most problems have a maximum route duration, while only 24% are constrained by time windows. The reason can be seen in the underlying application cases: waste collection, other municipal services, and most of the addressed logistics services (e.g., heating oil distribution) are



Figure 3 Types of problem variants for intermediate stops for replenishment and unloading.

Table 3	Types of	problem	variants	for	intermediate	stops	for re	plenishment	and	unloadin	g.
---------	----------	---------	----------	-----	--------------	-------	--------	-------------	-----	----------	----

AIF	Mourão and Almeida (2000), Ghiani, Improta, and Laporte (2001), Mourão and Amado (2005), Del Pia and Filippi (2006), Amaya, Langevin, and Trépanier (2007), Santos, Coutinho-Rodrigues, and Current (2008), Amaya, Langevin, and
	Trépanier (2010), Salazar-Aguilar, Langevin, and Laporte (2012, 2013), Belenguer et al. (2016)
NIF	Beltrami and Bodin (1974), Jordan (1987), Gerdessen (1996), Angelelli and Speranza (2002b), Chao (2002), Scheuerer (2006),
	Villegas et al. (2011), Bartolini and Schneider (2017)
NIFMRD	Bard et al. (1998), Crevier, Cordeau, and Laporte (2007), Kek, Cheu, and Meng (2008), Tarantilis, Zachariadis, and
	Kiranoudis (2008), Benjamin and Beasley (2010), Coene, Arnout, and Spieksma (2010), Buhrkal, Larsen, and Ropke
	(2012), Hemmelmayr et al. (2013a,b), Muter, Cordeau, and Laporte (2014), Schneider, Stenger, and Hof (2015), Willemse and Joubert (2016)
NIFTW	Kim, Kim, and Sahoo (2006), Benjamin and Beasley (2010), Villegas et al. (2010), Buhrkal, Larsen, and Ropke (2012), Drexl
	(2014), Prescott-Gagnon, Desaulniers, and Rousseau (2014), Markov, Varone, and Bierlaire (2016), Rothenbächer, Drexl, and Irnich (2016), Parragh and Cordeau (2017)
AIFMRD	De Rosa et al. (2002), Ghiani et al. (2004), Tan, Chew, and Lee (2006), Polacek et al. (2008), Ghiani et al. (2010)
AIFTW	Semet and Taillard (1993), Kim, Kim, and Sahoo (2006), Derigs, Pullmann, and Vogel (2013)

only limited by the daily planning horizon. Therefore, time windows only arise in specific application cases or in pure methodological contributions to challenge the algorithms.

Table 4 and Figure 4 provide a summary of the objectives. As can be seen, most publications focus on cost or distance minimization. Furthermore, the minimization of the overall route duration or of the number of vehicles is often considered. Only a few publications minimize the makespan of the longest tour in order to obtain tours of similar duration (cf. Salazar-Aguilar, Langevin, and Laporte 2012, 2013).

Table 5 details the solution approaches that have been used to solve RPIS for replenishment or unloading. To keep the table concise, we limited the solution methods listed in the heuristic section to algorithms that are used in more than one paper, and we merge certain algorithms in their respective class (e.g., two-phase algorithms contain route-first cluster-second algorithms). The majority are metaheuristics, which is mainly due to the problem size of most application cases (cf. Tables 6 and 7). Note that the listed MIPs are mostly used to provide a formal problem definition and not to solve the problem. Exact solution methods focus on the most promising approaches for routing problems (e.g., BP algorithms (Muter, Cordeau, and Laporte 2014, Parragh and Cordeau 2017, Prescott-Gagnon, Desaulniers, and Rousseau 2014)). A first BPC algorithm for the TTRP



Figure 4 Objectives for intermediate stops for replenishing and unloading.

Table 4	Objectives	for inter	mediate stops	for rep	plenishing	and	unloading.

Objective	References
min. costs	Mourão and Almeida (2000), Ghiani, Improta, and Laporte (2001), De Rosa et al. (2002), Ghiani et al. (2004, 2005), Mourão and Amado (2005), Amaya, Langevin, and Trépanier (2007), Crevier, Cordeau, and Laporte (2007), Kek,
	Cheu, and Meng (2008), Polacek et al. (2008), Amaya, Langevin, and Trépanier (2010), Ghiani et al. (2010),
	Buhrkal, Larsen, and Ropke (2012), Hemmelmayr et al. (2013a,b), Drexl (2014), Schneider, Stenger, and Hof (2015),
	Belenguer et al. (2016), Markov, Varone, and Bierlaire (2016), Rothenbächer, Drexl, and Irnich (2016), Willemse and
	Joubert (2016), Bartolini and Schneider (2017), Parragh and Cordeau (2017)
min. distance	Jordan (1987), Semet and Taillard (1993), Bard et al. (1998), Angelelli and Speranza (2002b,a), Chao (2002),
	Scheuerer (2006), Tan, Chew, and Lee (2006), Benjamin and Beasley (2010), Villegas et al. (2010, 2011), Derigs,
	Pullmann, and Vogel (2013), Prescott-Gagnon, Desaulniers, and Rousseau (2014)
min. duration	Beltrami and Bodin (1974), Gerdessen (1996), Del Pia and Filippi (2006), Kim, Kim, and Sahoo (2006), Santos,
	Coutinho-Rodrigues, and Current (2008), Tarantilis, Zachariadis, and Kiranoudis (2008), Coene, Arnout, and Spieksma (2010), Salazar-Aguilar, Langevin, and Laporte (2012, 2013), Muter, Cordeau, and Laporte (2014)
min. vehicles	Beltrami and Bodin (1974), Kim, Kim, and Sahoo (2006), Tan, Chew, and Lee (2006), Willemse and Joubert (2016)

was proposed by Rothenbächer, Drexl, and Irnich (2016). In addition, several metaheuristic algorithms incorporate dynamic programming components to optimally locate intermediate stops on

Table 5 Solution methods for routing problems on intermediate stops for replenishment and unloading.

	Exact					
(M)I(L)P	Jordan (1987), Bard et al. (1998), Mourão and Almeida (2000), Ghiani, Improta, and Laporte (2001), Chao (2002), Amaya, Langevin, and Trépanier (2007), Crevier, Cordeau, and Laporte (2007), Kek, Cheu, and Meng (2008), Amaya, Langevin, and Trépanier (2010), Coene, Arnout, and Spieksma (2010), Buhrkal, Larsen, and Ropke (2012), Hemmelmayr et al. (2013a,b), Drexl (2014), Schneider, Stenger, and Hof (2015), Belenguer et al. (2016), Markov, Varone, and Bierlaire (2016)					
BC	Bard et al. (1998), Amaya, Langevin, and Trépanier (2010), Drexl (2014), Belenguer et al. (2016), Bartolini and Schneider (2017)					
DP component	Hemmelmayr et al. (2013a), Schiffer et al. (2017b)					
BP, CG	Muter, Cordeau, and Laporte (2014), Prescott-Gagnon, Desaulniers, and Rousseau (2014), Parragh and Cordeau (2017)					
LB techniques	Mourão and Almeida (2000), Mourão and Amado (2005)					
BPC	Rothenbächer, Drexl, and Irnich (2016)					
	Heuristic					
TS	Semet and Taillard (1993), Angelelli and Speranza (2002b,a), Chao (2002), De Rosa et al. (2002), Ghiani et al. (2004), Scheuerer (2006), Crevier, Cordeau, and Laporte (2007), Tarantilis, Zachariadis, and Kiranoudis (2008), Benjamin and Beasley (2010). Prescott-Garon. Desaulniers, and Rousseau (2014)					
(A)LNS	Buhrkal, Larsen, and Ropke (2012), Salazar-Aguilar, Langevin, and Laporte (2012), Derigs, Pullmann, and Vogel (2013), Salazar-Aguilar, Langevin, and Laporte (2013), Schiffer et al. (2017b), Parragh and Cordeau (2017)					
(A)VNS	Del Pia and Filippi (2006), Polacek et al. (2008), Tarantilis, Zachariadis, and Kiranoudis (2008), Benjamin and Beasley (2010), Villegas et al. (2010, 2011), Hemmelmayr et al. (2013a,b), Prescott-Gagnon, Desaulniers, and Rousseau (2014), Schneider, Stenger, and Hof (2015), Markov, Varone, and Bierlaire (2016)					
LS	Gerdessen (1996), Del Pia and Filippi (2006), Tarantilis, Zachariadis, and Kiranoudis (2008), Derigs, Pullmann, and Vogel (2013), Hemmelmayr et al. (2013a), Markov, Varone, and Bierlaire (2016), Schiffer et al. (2017b)					
two-phase algorithm	Mourão and Almeida (2000), Ghiani et al. (2005), Kim, Kim, and Sahoo (2006), Coene, Arnout, and Spieksma (2010), Amaya, Langevin, and Trépanier (2010), Willemse and Joubert (2016)					
path-scanning	Santos, Coutinho-Rodrigues, and Current (2008), Willemse and Joubert (2016)					
EA	Tan, Chew, and Lee (2006), Ghiani et al. (2010)					
Other	Beltrami and Bodin (1974), Jordan (1987), Ghiani, Improta, and Laporte (2001), Mourão and Amado (2005), Crevier, Cordeau, and Laporte (2007), ?), Santos, Coutinho-Rodrigues, and Current (2008), Villegas et al. (2010, 2011)					

Abbreviations hold as follows: (M)I(L)P - (mixed) integer (linear) program; BC - branch-and-cut; DP - dynamic programing; BP - branch-and-price; CG - column generation, LB - lower bound, BPC - branch-and-price-and-cut; TS - tabu search; (A)LNS - (adaptive) large neighborhood search; (A)VNS - (adaptive) variable neighborhood search; LS - local search; EA - evolutionary algorithms.

routes (e.g., Hemmelmayr et al. 2013a, Schiffer et al. 2017b). As can be seen, TS and VNS are the most popular algorithms for this problem class. Contrary to other VRP variants, evolutionary algorithms, which turned out to be effective for a wide class of VRPs (cf. Vidal et al. 2012, 2013, 2014), are only rarely used.

Tables 6 and 7 provide an overview of the benchmark sets that have been published for node routing (Table 6) and arc routing (Table 7) with intermediate stops for replenishment and unloading. The tables show the number of instances (I), the number of nodes (N) or arcs (A), and the number of intra-route facilities (IF). As can be seen, some instance sets established as standard benchmark sets in the last years, while others have only been used by the authors themselves. For TTRPs, the benchmark set of Chao (2002) is the most used. For VRPs with replenishment stops, this role is taken by the benchmark sets of Crevier, Cordeau, and Laporte (2007) and by Tarantilis, Zachariadis, and Kiranoudis (2008). Large-scale instance were developed by Benjamin and Beasley (2010) and Kim, Kim, and Sahoo (2006) from the case studies on waste collection.

Concluding, RPIS for unloading and synchronization represent the majority of problems in the analyzed application area, while problems focusing on intermediate stops for replenishing in classical logistics applications account for a share of only 21%. Both arc routing and node routing problems have been solved. While the first are often related to municipal services, the latter arise mostly in classical distribution services. Most notably, uncertainties have not been considered so

Reference	Type	Ι	Ν	IF	Used within
Gerdessen (1996)	NIF	150	50 - 200	50 - 200	
Chao (2002)	NIF	21	50 - 199	13 - 150	Scheuerer (2006), Villegas et al. (2011), Derigs,
Angelelli and Speranza (2002b)	NIF	42	50-288	1 - 4	Pullmann, and Vogel (2013), Belenguer et al. (2016)
Kim, Kim, and Sahoo (2006)	NIFTW	10	102 - 2100	1 1	Benjamin and Beasley (2010), Buhrkal, Larsen, and
	NUENDE	0.0	10.000	0 -	Ropke $(2012)$
Crevier, Cordeau, and Laporte (2007)	NIFMRD	22	48-288	3-7	Tarantilis, Zachariadis, and Kiranoudis (2008), Hemmelmayr et al. (2013a), Muter, Cordeau, and Laporte (2014), Schneider, Stenger, and Hof (2015), Markov, Varone, and Bierlaire (2016), Schiffer et al. (2017b)
Tarantilis, Zachariadis, and Kiranoudis (2008)	NIFMRD	54	50 - 175	3-8	Hemmelmayr et al. (2013a), Schneider, Stenger, and Hof (2015), Schiffer et al. (2017b)
Drexl(2014)	NIFTW	109	16 - 106		Rothenbächer, Drexl, and Irnich (2016), Bartolini and Schneider (2017)
Prescott-Gagnon, Desaulniers, and Rousseau (2014)	NIFTW	18	250 - 750		
Willemse and Joubert (2016)	NIFMRD	3	1012 - 2755	2	
Parragh and Cordeau (2017)	NIFTW	18	25 - 100	7 - 75	Rothenbächer, Drexl, and Irnich (2016)
Case studies		2	184 - 387	1 - 3	Hemmelmayr et al. (2013b), Markov, Varone, and Bierlaire (2016)

Table 6 Instances for node routing problems with intermediate stops for replenishment or unloading.

If the number of intra-route facilities is not known, IF is left empty.

Table 7 Instances for arc routing problems with intermediate stops for replenishment or unloading.

Reference	Type	Ι	А	IF	Used within
Mourão and Almeida (2000)	AIF	30	13 - 97	1	
Ghiani, Improta, and Laporte (2001)	AIF	51	11 - 97	1 - 2	Ghiani et al. (2004), Polacek et al. (2008), Ghiani et al. (2010), De Rosa et al. (2002)
Mourão and Amado (2005)	AIF	30	94 - 743	1	
Amaya, Langevin, and Trépanier (2007)	AIF	180	50 - 595	$_{3-5}$	Amaya, Langevin, and Trépanier (2010)
Salazar-Aguilar, Langevin, and Laporte (2012)	AIF	45	113 - 795		
Salazar-Aguilar, Langevin, and Laporte (2013)	AIF	60	200 - 350	60 - 100	
Case studies		2	376 - 422		Ghiani et al. (2005), Del Pia and Filippi (2006)

If the number of intra-route facilities is not known, IF is left empty.

far. While neglecting uncertainties seems to be appropriate for some of the application cases (e.g., waste collection, road painting, small package shipping), considering uncertain demand in delivery problems with raw materials (e.g., propane distribution) or uncertain travel times in problem variants with time windows seems to constitute a promising research direction. Another promising research direction arises within the context of city logistics. Besides considering uncertainty, dynamic and stochastic problems arise in the context of e-commerce and same-day or express deliveries, and online algorithms are required to address these challenges.

### 3. Intermediate stops for refueling

We now consider publications that focus on intermediate stops for refueling. This type of intermediate stops arises in both dense and sparse refueling infrastructures. While dense infrastructures exist for ICEV fleets, sparse infrastructures arise mainly for AFV fleets. Section 3.1 focuses on refueling stops in dense refueling infrastructure. Intermediate stops for sparse refueling infrastructures are discussed in Section 3.2. Finally, Section 3.3 concludes this discussion with a detailed analysis of problem variants and algorithms.

#### 3.1. Intermediate stops for refueling in dense refueling networks

RPIS for refueling in dense refueling networks are relevant for ICEV fleets due to economic reasons, e.g., to take advantage of price differences at spatially close gas filling stations.

The first papers in this context were written by Ichimori and Ishii (1981) and Ichimori, Ishii, and Nishida (1983), focusing on a shortest path problem (SPP) for vehicles with limited fuel capacity and refueling options that are limited to dedicated nodes. The authors presented a modified Dijkstra algorithm to solve this problem. Bousonville et al. (2011) included refueling decisions for ICEVs into a VRPTW and analyzed the impact of price variations, especially on the tour length. The objective focused on minimizing the overall costs for refueling, and Solomon's I1 construction heuristic was applied. Khuller, Malekian, and Mestre (2011) studied SPPs and traveling salesman problems (TSPs) with price varying refueling options. Suzuki (2012) focused on a TSP with time windows and time-sensitive demand, considering refueling options. In addition, Suzuki and Dai (2012) proposed a variable reduction technique for this problem.

#### **3.2.** Intermediate stops for refueling in sparse refueling networks

Intermediate stops for refueling with sparse refueling structures arise mainly for ECVs and other AFVs. VRP variants, SPPs and TSPs, as well as location-routing problem (LRP) variants have been studied in this context.

Goncalves et al. (2011) considered a VRP with pickups and deliveries and a mixed fleet of ICEVs and ECVs to study the integration of ECVs in the fleet of a battery distributor. The authors presented a MIP minimizing fixed vehicle costs and routing costs. Although recharging time for intermediate stops is considered, dedicated charging station vertices are not used. Conrad and Figliozzi (2011) introduced the recharging VRP in which vehicles with a limited driving range are allowed to recharge en route at certain customer locations, while considering a fixed recharging time and customer time windows. The authors used a lexicographic objective function to first minimize the number of vehicles and then the routing cost. Erdoğan and Miller-Hooks (2012) proposed the green VRP (GVRP) that considers a limited fuel capacity for AFVs and refueling options on routes, while restricting the maximum duration of a route. The authors proposed a modified savings algorithm (cf. Clarke and Wright 1964) and a density based route-first cluster-second algorithm. Focusing on an airport shuttle service, Barco et al. (2013) presented a comprehensive approach for integrating ECVs into a fleet of shuttle vehicles. Schneider, Stenger, and Goeke (2014) were the first to address the electric VRP (EVRP) with time windows (EVRPTW) focusing on a pure electric vehicle fleet and dedicated vertices for recharging activities considering quantity-dependent recharging times. The authors also used the lexicographic objective function of Conrad and Figliozzi (2011).

Felipe et al. (2014) introduced the GVRP with multiple technologies and partial recharges (GVRPMTPR), focusing on different types of recharging stations for ECV and taking different costs, different charging speeds, and partial recharging into consideration. Sassi, Cherif-Khettaf, and Oulamara (2015c) presented an EVRP with partial recharging and a heterogeneous ECV fleet, with a lexicographic objective function that first minimizes the number of vehicles and then the distance and charging cost. The same authors presented a multi-start iterated local search in Sassi, Cherif-Khettaf, and Oulamara (2015b) and an iterated TS in Sassi, Cherif-Khettaf, and Oulamara (2015a) for this problem. Goeke and Schneider (2015) analyzed an EVRP with a mixed fleet of ICEVs and ECVs, considering a realistic energy consumption function using data on vehicle speed, vehicle load, and gradients. Bruglieri et al. (2015a) and Bruglieri et al. (2015b) proposed a matheuristic based on a variable neighborhood branching for the EVRPTW. Montoya et al. (2016) developed a multi-space sampling heuristic for the GVRP. Verma, Lamsal, and Keough (2015) investigated the EVRPTW for battery swapping stations instead of recharging stations. The authors developed a VNS to solve this problem and calculated results on the instances of Schneider, Stenger, and Goeke (2014). Hiermann et al. (2016) investigated the EVRPTW with heterogeneous electric vehicles that have different acquisition costs and vehicle-independent routing costs.

Desaulniers et al. (2016) developed a BPC algorithm for the EVRPTW, covering four variants with single and multiple recharge stops per route as well as full and partial recharging. The authors presented a mono-directional and a bi-directional pricing labeling algorithm and found that multiple recharges improve the overall solution with respect to the number of vehicles and costs. Keskin and Catay (2016) addressed the EVRPTW and partial recharging (EVRPTWPR) and developed an ALNS. Koç and Karaoglan (2016) focused on the GVRP and introduced a BC algorithm to improve lower bounds and a simulated annealing (SA) algorithm to calculate upper bounds. Yavuz and Capar (2017) discussed the adoption of AFVs in service fleets. Montoya et al. (2017) investigated an EVRP allowing for partial recharging and considering a non-linear charging function. The authors presented a hybrid algorithm based on VNS and LS. Additionally, they presented a component to insert charging stations into routes, either based on a greedy heuristic or on a mixed integer program. Yavuz (2017) proposed an iterated beam search algorithm for the GVRP. Andelmin and Bartolini (2017) presented a column generation based approach to solve the GVRP exactly up to 111 customers. Schiffer et al. (2017a) introduced the EVRP with truck driver scheduling (EVRPTDS) and analyzed the impact of HOS regulations on the competitiveness of ECVs compared to ICEVs, synchronizing idle times for recharging and breaking. The authors presented an ALNS-based algorithm with a time-efficient HOS scheduling component and analyzed European Union (EU) as well as United States (US) HOS regulations. Froger et al. (2017a) introduced an alternative formulation for the EVRP based on a multigraph and Froger et al. (2017b) focused on capacitated charging stations.

The first publications on generic VRPIS variants only appeared recently. Schneider, Stenger, and Hof (2015) developed a generalized VRPIS model and presented an adaptive VNS that provided good results on the GVRP and EVRP variants with full recharging options. Schiffer et al. (2017b) developed a generic algorithmic framework for VRPISs based on an ALNS with an additional DP element. This algorithm yields the best known results for several EVRP variants, namely the EVRPTW, the EVRPTWPR, and the EVRP with maximum route duration, and the GVRP.

Besides publications on VRP variants, SPPs and TSPs have been investigated. Liao, Lu, and Shen (2016) introduced the electric vehicle touring problem that accounts for a shortest route that can be chosen by an ECV to get from an origin to a destination. On this route, the ECV may stop at one or several battery swapping stations to switch its battery. The authors presented a polynomialtime algorithm for this problem. Roberti and Wen (2016) introduced the electric TSP with time windows and presented a three-phase heuristic based on a VNS and dynamic programming to solve instances with up to 200 customers. Further work on routing pure ECVs has been presented by Sweda, Dolinskaya, and Klabjan (2017), accounting for uncertain recharging options over time, so that adaptive routing and recharging decisions can be optimized.

The first papers focusing on hybrid electric vehicles (HEVs) were published only recently. Arslan, Yildiz, and Karasan (2015) presented the minimum cost path problem for plug-in hybrid electric vehicles, considering refueling and recharging stations with different cost structures. The authors presented a dynamic programming and a shortest path algorithm, minimizing overall costs. Doppstadt, Koberstein, and Vigo (2016) introduced the hybrid electric vehicle TSP in which vehicles can switch between different engine modes and presented a TS heuristic to solve large-sized instances. Another publication on hybrid electric vehicles was presented by Mancini (2017), who introduced the hybrid VRP in which vehicles can either switch their engine mode once the battery is discharged or can be recharged at specific charging stations. Nejad et al. (2017) focused on optimal routing for plug-in hybrid electric vehicles and provided different exact DP-based algorithms for this problem.

The first publications on LRPs in the context of ECVs were also recently published. Because decisions on vehicle routing and charging station locations are interdependent, a simultaneous consideration bears a significant improvement potential at strategic level (cf. Schiffer and Walther 2017b). These publications consider variants of the LRP with intra-route facilities (LRPIF) (cf. Schiffer and Walther 2017a). The LRPIF differs from conventional LRPs because the decision is on locating intra-route facilities as introduced in Section 1 instead of depots. In the following,

we focus on LRPIF variants and refer to recent surveys (Lopes et al. 2013, Drexl and Schneider 2014, Prodhon and Prins 2014, Cuda, Guastaroba, and Speranza 2015, Albareda-Sambola 2015, Schneider and Drexl 2017) for an overview of conventional LRPs.

The first LRP in this context is by Yang and Sun (2015) who introduced the battery swap station electric vehicle LRP (BSS-EV-LRP) that simultaneously determines battery swapping station locations and vehicle routes. Hof, Schneider, and Goeke (2017) extended the VNS developed in Schneider, Stenger, and Hof (2015) to the BSS-EV-LRP and significantly improved the results of Yang and Sun (2015). Schiffer and Walther (2017b) introduced the electric LRP with time windows and partial recharging (ELRPTWPR), which extends the BSS-EV-LRP to a more general problem formulation, accounting for partial recharging, time windows and time-dependent recharging. The authors discussed different objective functions and highlighted the impact of simultaneous charging station location and vehicle routing decisions. Schiffer, Stütz, and Walther (2016) presented a case study for the ELRPTWPR based on the distribution network of a German retail company and showed that ECVs are on the verge of breaking even for certain application cases. Since customer patterns heavily affect the routing and the interdependent charging station location decision, Schiffer and Walther (2017c) introduced a robust ELRPTWPR that considers uncertainty in customer patterns with regard to the spatial distribution, demand and time windows. The authors presented a parallelized ALNS to solve this problem. In a more generic fashion, Schiffer and Walther (2017a) introduced the LRPIF which is not limited to ECVs and charging stations but also accounts for conventional vehicles or other AFVs and intra-route facilities for freight replenishment. The authors presented new benchmark instances and an ALNS with a DP component. Schiffer, Schneider, and Laporte (2017) extended the LRPIF for combined facilities at which recharging energy and replenishing freight can take place simultaneously. Furthermore, the authors integrated lower bounding techniques to avoid unpromising facility configurations in an ALNS. This algorithm yields the best results for all LRPIF variants discussed above.

#### 3.3. Analysis of intermediate stops for refueling

Table 8 summarizes the scope of the publications discussed in this section. Besides i) the type of publication and ii) the type of the routing problem, which have also been discussed for Section 2, we include additional characteristics. We further differentiate between iii) the vehicle type, iv) the type of data (as in Section 2), v) the modeling approach, and vi) the refueling infrastructure. As can be seen in Figure 5, the share of publications that describe case studies is rather low (15%), while most publications focus on methodological improvements (85%). Problems on intermediate stops for refueling are mostly tackled as pure routing problems (85%) (TSP/SPP, VRP); a much smaller proportion (15%) also considers the location component. All publications consider node

	Table 0 Scope of publications of intermediate stops for refueling.
Case Study Methodological	<ul> <li>Gonçalves et al. (2011), Barco et al. (2013), Sassi, Cherif-Khettaf, and Oulamara (2015c), Schiffer, Stütz, and Walther (2016), Yavuz and Çapar (2017), Nejad et al. (2017), Schiffer et al. (2017a)</li> <li>Ichimori and Ishii (1981), Ichimori, Ishii, and Nishida (1983), Bousonville et al. (2011), Conrad and Figliozzi (2011), Khuller, Malekian, and Mestre (2011), Erdoğan and Miller-Hooks (2012), Suzuki (2012), Felipe et al. (2014), Hiermann et al. (2016), Schneider, Stenger, and Goeke (2014), Arslan, Yildiz, and Karasan (2015), Bruglieri et al. (2015b,a), Goeke and Schneider (2015), Mancini (2017), Montoya et al. (2016, 2017), Sassi, Cherif-Khettaf, and Oulamara (2015a,b), Schiffer and Walther (2017b), Schneider, Stenger, and Hof (2015), Verma, Lamsal, and Keough (2015), Yang and Sun (2015), Desaulniers et al. (2016), Andelmin and Bartolini (2017), Froger et al. (2017a,b), Hof, Schneider, and Goeke (2017), Schiffer and Walther (2017a), Schiffer et al. (2017b), Schiffer and Walther (2017c), Schiffer et al. (2017b), Schiffer and Walther (2017c), Schiffer, Schneider, and Laporte (2017), Sweda, Dolinskaya, and Klabjan (2017), Yavuz (2017)</li> </ul>
Node routing	<ul> <li>Ichimori and Ishii (1981), Ichimori, Ishii, and Nishida (1983), Bousonville et al. (2011), Conrad and Figliozzi (2011), Gonçalves et al. (2011), Khuller, Malekian, and Mestre (2011), Erdoğan and Miller-Hooks (2012), Suzuki (2012), Barco et al. (2013), Felipe et al. (2014), Hiermann et al. (2016), Schneider, Stenger, and Goeke (2014), Arslan, Yildiz, and Karasan (2015), Bruglieri et al. (2015b,a), Goeke and Schneider (2015), Mancini (2017), Montoya et al. (2016, 2017), Sassi, Cherif-Khettaf, and Oulamara (2015c,a,b), Schiffer and Walther (2017b), Schneider, Stenger, and Hof (2015), Verma, Lamsal, and Keough (2015), Yang and Sun (2015), Desaulniers et al. (2016), Doppstadt, Koberstein, and Vigo (2016), Keskin and Çatay (2016), Koç and Karaoglan (2016), Roberti and Wen (2016), Schiffer, Stütz, and Walther (2016), Andelmin and Bartolini (2017), Froger et al. (2017a,b), Hof, Schneider, and Goeke (2017), Nejad et al. (2017), Schiffer and Walther (2017a), Schiffer et al. (2017b), Schiffer and Walther (2017c), Schiffer, Schneider, and Laporte (2017), Schiffer et al. (2017a), Sweda, Dolinskaya, and Klabjan (2017), Yavuz and Çapar (2017), Yavuz (2017)</li> </ul>
HEV ICEV	<ul> <li>Conrad and Figliozzi (2011), Gonçalves et al. (2011), Barco et al. (2013), Felipe et al. (2014), Hiermann et al. (2016), Schneider, Stenger, and Goeke (2014), Bruglieri et al. (2015b,a), Goeke and Schneider (2015), Montoya et al. (2017), Sassi, Cherif-Khettaf, and Oulamara (2015c,a,b), Schiffer and Walther (2017b), Schneider, Stenger, and Hof (2015), Verma, Lamsal, and Keough (2015), Yang and Sun (2015), Desaulniers et al. (2016), Keskin and Çatay (2016), Koç and Karaoglan (2016), Roberti and Wen (2016), Schiffer, Stütz, and Walther (2016), Froger et al. (2017a,b), Hof, Schneider, and Goeke (2017), Schiffer and Walther (2017a), Schiffer et al. (2017b), Schiffer and Walther (2017c), Schiffer, and Laporte (2017), Schiffer et al. (2017a), Sweda, Dolinskaya, and Klabjan (2017)</li> <li>Arslan, Yildiz, and Karasan (2015), Mancini (2017), Doppstadt, Koberstein, and Vigo (2016), Nejad et al. (2017)</li> <li>Ichimori and Ishii (1981), Ichimori, Ishii, and Nishida (1983), Bousonville et al. (2011), Khuller, Malekian, and Mestre (2017), Scrubier (2012)</li> </ul>
AFV	(2011), SUZURI (2012) Erdoğan and Miller-Hooks (2012), Schneider, Stenger, and Hof (2015), Montoya et al. (2016), Andelmin and Bartolini (2017), Schiffer et al. (2017b), Yavuz and Çapar (2017), Yavuz (2017)
Deterministic	<ul> <li>Bousonville et al. (2011), Conrad and Figliozzi (2011), Gonçalves et al. (2011), Khuller, Malekian, and Mestre (2011), Erdoğan and Miller-Hooks (2012), Suzuki (2012), Barco et al. (2013), Felipe et al. (2014), Hiermann et al. (2016), Schneider, Stenger, and Goeke (2014), Bruglieri et al. (2015b,a), Goeke and Schneider (2015), Mancini (2017), Montoya et al. (2016, 2017), Sassi, Cherif-Khettaf, and Oulamara (2015c,a,b), Schiffer and Walther (2017b), Verma, Lamsal, and Keough (2015), Yang and Sun (2015), Desaulniers et al. (2016), Doppstadt, Koberstein, and Vigo (2016), Keskin and Çatay (2016), Koç and Karaoglan (2016), Roberti and Wen (2016), Schiffer, Stütz, and Walther (2016), Andelmin and Bartolini (2017), Froger et al. (2017a,b), Hof, Schneider, and Goeke (2017), Nejad et al. (2017), Schiffer and Walther (2017a), Schiffer et al. (2017b), Schiffer, Schneider, and Laporte (2017), Schiffer et al. (2017a), Yavuz and Çapar (2017), Yavuz (2017)</li> <li>Sweda, Dolinskaya, and Klabian (2017), Schiffer and Walther (2017c)</li> </ul>
TSP / SPP	Ichimori and Ishii (1981), Ichimori, Ishii, and Nishida (1983), Khuller, Malekian, and Mestre (2011), Suzuki (2012),
VRP	<ul> <li>Arslan, Yildiz, and Karasan (2015), Doppstadt, Koberstein, and Vigo (2016), Roberti and Wen (2016), Sweda, Dolinskaya, and Klabjan (2017), Nejad et al. (2017)</li> <li>Bousonville et al. (2011), Conrad and Figliozzi (2011), Gonçalves et al. (2011), Erdoğan and Miller-Hooks (2012), Barco et al. (2013), Felipe et al. (2014), Hiermann et al. (2016), Schneider, Stenger, and Goeke (2014), Bruglieri et al. (2015b,a), Goeke and Schneider (2015), Mancini (2017), Montoya et al. (2016, 2017), Sassi, Cherif-Khettaf, and Oulamara (2015c,a,b), Schneider, Stenger, and Hof (2015), Verma, Lamsal, and Keough (2015), Desaulniers et al. (2016), Keskin and Çatay (2016), Koç and Karaoglan (2016), Andelmin and Bartolini (2017), Froger et al. (2017a,b), Schiffer et al. (2017b), Yavuz and Çapar (2017), Schiffer et al. (2017a).</li> </ul>
LRP	Schiffer and Walther (2017b), Yang and Sun (2015), Schiffer and Walther (2017a), Schiffer, Stütz, and Walther (2016), Hof, Schneider, and Goeke (2017), Schiffer and Walther (2017c), Schiffer, Schneider, and Laporte (2017)
Dense	Ichimori and Ishii (1981), Ichimori, Ishii, and Nishida (1983), Bousonville et al. (2011), Khuller, Malekian, and Mestre (2011), Suzuki (2012)
Sparse	<ul> <li>Gonçalves et al. (2011), Conrad and Figliozzi (2011), Erdoğan and Miller-Hooks (2012), Barco et al. (2013), Schneider, Stenger, and Goeke (2014), Felipe et al. (2014), Arslan, Yildiz, and Karasan (2015), Goeke and Schneider (2015), Bruglieri et al. (2015b,a), Sassi, Cherif-Khettaf, and Oulamara (2015a,bc,), Schneider, Stenger, and Hof (2015), Verma, Lamsal, and Keough (2015), Yang and Sun (2015), Hiermann et al. (2016), Desaulniers et al. (2016), Doppstadt, Koberstein, and Vigo (2016), Keskin and Çatay (2016), Koç and Karaoglan (2016), Montoya et al. (2016), Roberti and Wen (2016), Andelmin and Bartolini (2017), Froger et al. (2017a,b), Schiffer, Stütz, and Walther (2016), Mancini (2017), Montoya et al. (2017), Nejad et al. (2017), Schiffer and Walther (2017b,a), Hof, Schneider, and Goeke (2017), Schiffer and Walther (2017c), Schiffer and Goeke (2017), Schiffer et al. (2017a,b), Sweda, Dolinskaya, and Klabjan (2017), Yavuz and Çapar (2017), Yavuz (2017)</li> </ul>

Scope of publications on intermediate stops for refueling

routing problems. The large majority of publications focuses on ECVs (67%), while only a limited number of publications focus on HEVs (8%) or on AFVs (15%) (cf. Figure 6). Only 10% consider ICEVs. This corresponds to the proportion of problems in which a dense refueling structure is considered (11%), while all publications that are related to any kind of AFVs consider a sparse refueling structure.

Table 8



Figure 5 Characteristics of publications on intermediate stops for refueling.



Figure 6 Characteristics of publications on intermediate stops for refueling.

Figure 7 and Table 9 detail the problem types, based on the definition given in Table 1. Because all problems focus on fuel or energy that is consumed while driving, the resource consumption is always arc-based. Considering the time restrictions, all possible characteristics are addressed. The majority of the problems considers time-dependent replenishment processes (57%), and time window restrictions are also considered (49%). Overall, 79% of the problems include time constraints. Furthermore, the models analyzed in this section are the only ones that account for partial replenishment because refueling consumes a significant amount of time for ECVs.

Figure 8 and Table 10 illustrate the different objective functions. As can be seen, most publications minimize overall costs or the total distance driven. Some publications make use of the lexicographic objective function approach used in heuristics on the classical VRPTW, minimizing the number of vehicles first and the total traveled distance second. Other objectives, e.g., minimizing the overall duration or consumed energy are only rarely applied.

Table 11 shows the different algorithms that have been used to solve routing problems with refueling stops. A majority of the problems use a MIP to define the analyzed problem in a formal way but not to create solutions on large-sized instances. Again, metaheuristics are more often used than exact algorithms. The few available exact algorithms are based on BP, BC, and BPC. Only for



Figure 7 Types of problem variants for intermediate stops for refueling.

Table 9 Types of problem variants for intermediate stops for refueling.

AIF	Ichimori and Ishii (1981), Ichimori, Ishii, and Nishida (1983), Khuller, Malekian, and Mestre (2011), Arslan, Yildiz, and Karasan (2015), Yang and Sun (2015), Doppstadt, Koberstein, and Vigo (2016), Liao, Lu, and Shen (2016), Hof, Schneider, and Goeke (2017), Nejad et al. (2017)
AIFMRD	Gonçalves et al. (2011), Erdoğan and Miller-Hooks (2012), Schneider, Stenger, and Hof (2015), Montoya et al. (2016), Koç and Karaoglan (2016), Andelmin and Bartolini (2017), Yavuz (2017)
AIFTW	Bousonville et al. (2011), Conrad and Figliozzi (2011), Suzuki (2012), Barco et al. (2013), Verma, Lamsal, and Keough (2015)
ADFMRD	Mancini (2017), Schneider, Stenger, and Hof (2015)
ADFTW	Hiermann et al. (2016), Schneider, Stenger, and Goeke (2014), Bruglieri et al. (2015a,b), Goeke and Schneider (2015), Schneider, Stenger, and Hof (2015), Desaulniers et al. (2016)
ADP	Sweda, Dolinskaya, and Klabjan (2017)
ADPMRD	Felipe et al. (2014), Montoya et al. (2017), Sassi, Cherif-Khettaf, and Oulamara (2015c,a,b), Yavuz and Çapar (2017)
ADPTW	Desaulniers et al. (2016), Keskin and Çatay (2016), Roberti and Wen (2016), Schiffer, Stütz, and Walther (2016), Froger et al. (2017a,b), Schiffer and Walther (2017b,a)
ADPIFTW	Schiffer, Schneider, and Laporte (2017), Schiffer et al. (2017a)





Table 10	Objectives	for intermed	liate stops f	for refuel	ing
----------	------------	--------------	---------------	------------	-----

Objective	References
Costs	Bousonville et al. (2011), Conrad and Figliozzi (2011), Gonçalves et al. (2011), Khuller, Malekian, and Mestre (2011), Suzuki (2012), Felipe et al. (2014), Hiermann et al. (2016), Arslan, Yildiz, and Karasan (2015), Goeke and Schneider (2015), Sassi, Cherif-Khettaf, and Oulamara (2015c,a,b), Schiffer and Walther (2017b), Verma, Lamsal, and Keough (2015), Yang and Sun (2015), Desaulniers et al. (2016), Doppstadt, Koberstein, and Vigo (2016), Schiffer and Walther (2017a), Schiffer, Stütz, and Walther (2016), Sweda, Dolinskaya, and Klabjan (2017), Yavuz and Çapar (2017), Hof, Schneider, and Goeke (2017), Schiffer, Schneider, and Laporte (2017), Schiffer et al. (2017a)
Distance	Ichimori and Ishii (1981), Ichimori, Ishii, and Nishida (1983), Erdoğan and Miller-Hooks (2012), Schneider, Stenger, and Goeke (2014), Bruglieri et al. (2015b,a), Montoya et al. (2016), Schneider, Stenger, and Hof (2015), Keskin and Çatay (2016), Koç and Karaoglan (2016), Liao, Lu, and Shen (2016), Roberti and Wen (2016), Andelmin and Bartolini (2017), Mancini (2017), Schiffer and Walther (2017b), Yavuz and Capar (2017), Yavuz (2017)
duration	Froger et al. (2017a,b), Montoya et al. (2017)
Num. Veh. / trips	Conrad and Figliozzi (2011), Schneider, Stenger, and Goeke (2014), Bruglieri et al. (2015b,a), Sassi, Cherif-Khettaf, and Oulamara (2015c,a,b), Schiffer and Walther (2017b), Schneider, Stenger, and Hof (2015), Keskin and Çatay (2016)
Energy	Barco et al. (2013), Nejad et al. (2017)
Emissions	Yavuz and Çapar (2017)
Num stations	Schiffer and Walther (2017b)

-

	Exact
(M)I(L)P	Conrad and Figliozzi (2011), Gonçalves et al. (2011), Erdoğan and Miller-Hooks (2012), Felipe et al. (2014), Schneider, Stenger, and Goeke (2014), Bruglieri et al. (2015b,a), Goeke and Schneider (2015), Mancini (2017), Montoya et al. (2017), Sassi, Cherif-Khettaf, and Oulamara (2015c), Schiffer and Walther (2017b), Schneider, Stenger, and Hof (2015), Doppstadt, Koberstein, and Vigo (2016), Liao, Lu, and Shen (2016), Schiffer, Stütz, and Walther (2016), Schiffer and Walther (2017c)
B&P / CG	Hiermann et al. (2016), Montova et al. (2016), Andelmin and Bartolini (2017)
DP	Arslan, Yildiz, and Karasan (2015), Roberti and Wen (2016), Schiffer and Walther (2017a), Schiffer et al. (2017b), Schiffer, Stütz, and Walther (2016), Nejad et al. (2017), Schiffer and Walther (2017c)
B&C	Koç and Karaoglan (2016)
B&P&C	Desaulniers et al. (2016)
Other	Ichimori and Ishii (1981), Ichimori, Ishii, and Nishida (1983), Khuller, Malekian, and Mestre (2011), Liao, Lu, and Shen (2016), Sweda, Dolinskaya, and Klabjan (2017), Yavuz (2017)
	Heuristic
TS	Schneider, Stenger, and Goeke (2014), Sassi, Cherif-Khettaf, and Oulamara (2015a), Doppstadt, Koberstein, and Vigo (2016)
(A)LNS	Hiermann et al. (2016), Goeke and Schneider (2015), Mancini (2017), Yang and Sun (2015), Keskin and Çatay (2016), Schiffer and Walther (2017a), Schiffer et al. (2017b), Schiffer, Stütz, and Walther (2016), Schiffer and Walther (2017c), Schiffer, Schneider, and Laporte (2017), Schiffer et al. (2017a)
(A)VNS	Schneider, Stenger, and Goeke (2014), Bruglieri et al. (2015b,a), Montoya et al. (2017), Schneider, Stenger, and Hof (2015), Verma, Lamsal, and Keough (2015), Roberti and Wen (2016), Yavuz and Çapar (2017), Hof, Schneider, and Goeke (2017)
LS	Felipe et al. (2014), Hiermann et al. (2016), Goeke and Schneider (2015), Montoya et al. (2017), Sassi, Cherif-Khettaf, and Oulamara (2015c,b), Schneider, Stenger, and Hof (2015), Verma, Lamsal, and Keough (2015), Schiffer and Walther (2017a), Schiffer et al. (2017b), Schiffer, Stütz, and Walther (2016), Hof, Schneider, and Goeke (2017), Schiffer and Walther (2017c), Schiffer, Schneider, and Laporte (2017), Schiffer et al. (2017a)
SA	Suzuki (2012), Felipe et al. (2014), Goeke and Schneider (2015), Koç and Karaoglan (2016)
two-phase algorithm	Erdoğan and Miller-Hooks (2012), Montoya et al. (2016)
other	Bousonville et al. (2011), Conrad and Figliozzi (2011), Barco et al. (2013), Arslan, Yildiz, and Karasan (2015), Sweda, Dolinskaya, and Klabjan (2017)

Table 11 Solution methods.

Abbreviations hold as follows: (M)I(L)P - (mixed) integer (linear) program; B&C - branch-and-cut; DP - dynamic programing; B&P - branch-and-price; CG - column generation, LB - lower bound, B&P&C - branch-and-price-and-cut; TS - tabu search; (A)LNS - (adaptive) large neighborhood search; (A)VNS - (adaptive) variable neighborhood search; LS - local search; SA - simulated annealing.

SPPs, a few polynomial time algorithms have been presented (e.g., Sweda, Dolinskaya, and Klabjan (2017)). Metaheuristics mainly use ALNS and VNS. Contrary to problems with intermediate stops for unloading or replenishment, TS is only rarely used.

Table 12 summarizes the instance sets published so far. Three benchmark sets are used regularly to assess the competitiveness of algorithms. For the GVRP, there are the instance sets of Erdoğan and Miller-Hooks (2012), while the instance sets of Schneider, Stenger, and Goeke (2014) are used for EVRP variants. In a location-routing context, the instance set of Yang and Sun (2015) is used to assess the competitiveness of LRPIF algorithms.

Concluding, most publications on intermediate stops for refueling focus on recharging stops for ECVs, which is a highly relevant topic with benefits for sustainable transport developments, e.g., by reducing range anxiety concerns (Pelletier et al. 2017). Most contributions focus on algorithmic aspects or extend existing EVRP variants by additional real-world constraints (e.g., realistic energy consumption, non-linear recharging). However, publications on case studies and real-world problems are quite sparse, which may be due to the fact that the adoption rate of ECVs is still quite low, and the expected market uptake is slow due to the concerns of practitioners. Thus, case studies that help to highlight the competitiveness of ECVs (e.g., Schiffer, Stütz, and Walther 2016) and to reduce the concerns of practitioners on applicability or range anxiety seem to be an important research direction that, besides scientific contribution, adds societal value by boosting the market

reference	Type	Ι	Ν	IF	Used within
Conrad and Figliozzi (2011)	AIF	30	40		
Bousonville et al. (2011)	AIFTW	56	100	121 - 441	
Suzuki (2012)	AIFTW	6	10 - 20	10 - 20	
Erdoğan and Miller-Hooks (2012)	AIFMRD	52	20-500	21-28	Felipe et al. (2014), Mancini (2017), Montoya et al. (2016), Koç and Karaoglan (2016), Andelmin and Bartolini (2017), Schiffer et al. (2017b), Yavuz and Çapar (2017), Mancini (2017)
Felipe et al. (2014)	ADPMRD	60	100 - 400	5 - 9	
Schneider, Stenger, and Goeke (2014)	ADFTW	92	5-100	21	Felipe et al. (2014), Hiermann et al. (2016), Bruglieri et al. (2015a,b), Schiffer and Walther (2017b), Verma, Lamsal, and Keough (2015), Desaulniers et al. (2016), Keskin and Çatay (2016), Schiffer and Walther (2017a), Schiffer et al. (2017b)
Schneider, Stenger, and Hof (2015)	ADFMRD	34	50 - 480	1 - 20	Schiffer et al. (2017b)
Doppstadt, Koberstein, and Vigo (2016)	AIF	36	8-50		
Roberti and Wen (2016)	ADPTW	100	20 - 200	5 - 10	
Montoya et al. (2017)	ADPMRD	120	10 - 320	2 - 38	
Yang and Sun (2015)	AIF	24	16 - 480	det	Schiffer and Walther (2017a), Hof, Schneider, and Goeke (2017), Schiffer, Schneider, and Laporte (2017)
Schiffer and Walther (2017c)	ADPTW	90	100	det	
Schiffer and Walther (2017a)	ADPTW	24	18 - 160	det	
Schiffer et al. (2017a)	ADPIFTW	56	100	897	
Schiffer, Schneider, and Laporte (2017)	ADPIFTW	56	100	det	

Table 12 Instance sets for routing problems with intermediate stops for refueling.

If the number of intra-route facilities is not known or determined in an LRPIF, IF is left empty.

uptake of ECVs. In addition, alternative charging technologies constitute an interesting avenue for future research. Recharging lanes that help to recharge the battery while the vehicle is driving are currently tested for medium-duty ECVs in the logistics sector. A scientific evaluation of this concept and its implications on routing ECVs is still missing. Lastly, the consideration of uncertainties with respect to charging station availabilities should be considered in future work to generate robust route plans for ECV fleets.

#### 4. Intermediate stops for rests and breaks

The third field in which intermediate stops are required are routing problems that consider break and rest periods arising from HOS regulations or multi-day planning problems. The relevant literature can be classified into truck driver scheduling problems (TDSPs) that account for HOS regulations to which logistics fleets have to abide, and into team orienteering problems (TOPs) and TSPs with hotel selection that originated from trip planning in tourism. TDSPs can be further separated into classical TDSPs in which route plans are already fixed and only a break and rest sequence have to be scheduled on the routes, and vehicle routing and truck driver scheduling problems (VRTDSPs), in which route plans as well as break schedules are determined. Due to prefixed route plans, TDSPs do not fully match the scope of our survey. Section 4.1 gives an overview of VRTDSPs, and Section 4.2 focuses on TOPs and TSPs with hotel selection. In Section 4.3, we again analyze the characteristics of the considered publications.

#### 4.1. Vehicle routing problems with truck driver scheduling

The first publication of a VRP focusing on HOS is by Xu et al. (2003). The authors investigated a pickup and delivery VRP minimizing a cost objective that contains fixed, mileage and layover costs and paid special attention to additional real-world constraints, e.g., driver work rules. A CG based heuristic and lower bounding procedures were proposed to solve the problem. Ceselli, Righini, and Salani (2009) investigated another rich VRP with driver work rules, time windows and additional customer and freight restrictions. The authors presented a BP algorithm with a bidirectional labeling that solves the underlying pricing problem as an elementary shortest path problem. Goel (2009) introduced the VRTDSP by extending the standard VRPTW to HOS regulations. This work focused on the EU HOS regulations and applied a LNS heuristic. The author minimized the number of vehicles as first and the overall traveled distance as a secondary objective. Benchmark instances were created based on the VRPTW instances of Solomon (1987). Further work on integrating EU HOS regulations into the VRPTW was published by Kok et al. (2010). Besides basic HOS regulations that have already been addressed in Goel (2009), the authors considered additional regulations that allow for more flexibility by adding small exceptions to the daily driving time. A restricted DP heuristic was presented and was shown to outperform the algorithm of Goel (2009). The authors found that slight modifications of HOS rules yield a significant decrease for both, the number of vehicles and the driven distance. Another contribution on the VRPTW with EU HOS regulations was published by Prescott-Gagnon et al. (2010). The authors presented an LNS-based CG heuristic that clearly outperformed the algorithms of Goel (2009) and Kok et al. (2010). Kok, Hans, and Schutten (2011) developed a sequential insertion heuristic, focusing on the VRPTW with HOS regulations minimizing the route duration to keep some flexibility in case of traffic congestion. Results are discussed for a real-world case as well as for the Solomon benchmark instances. Rancourt, Cordeau, and Laporte (2013) focused on US HOS regulations, considering a heterogeneous fleet and multiple time windows. The authors developed a unified TS algorithm with heuristic scheduling approaches for assigning breaks. Besides benchmark instances based on the Solomon instances, the authors analyzed a real-world case. An algorithmic framework based on a hybrid genetic search was proposed by Goel and Vidal (2014), considering EU, US, Australian and Canadian HOS regulations. The authors investigated the impact of different HOS with respect to safety and economic efficiency in this context. The first exact algorithm for the VRTDSP was proposed by Goel and Irnich (2016), introducing a BP algorithm for EU and US HOS regulations. A bidirectional dynamic programming approach was applied to solve the pricing problem as an elementary shortest path problem. Koç, Jabali, and Laporte (2017) introduced the VRTDSP with idling options (VRTDSPIO), considering idling costs beside routing and driver costs. The authors presented a matheuristic combining ALNS with a MIP and showed results on the Solomon benchmark sets. Schiffer et al. (2017a) introduced the EVRPTDS and analyzed the impact of HOS on the competitiveness of ECVs compared to ICEVs. The authors focused on EU as well as US HOS regulations and presented an ALNS as well as new real-world based benchmark instances.

#### 4.2. Orienteering and traveling salesman problems with hotel selection

The TOP is also known as the multi-vehicle version of the selective TSP (cf. Laporte and Martello 1990). Within the TOP, a circle of maximum profit has to be determined on a weighted graph with profits associated with vertices, while this circle is not allowed to exceed a maximum distance or duration. The TOP is often applied to determine tourist trips or for traveling salespersons with limited time budgets. If multi-day trips are considered, hotel selection arises within TOPs as an idling variant. In the following, we analyze TOPs and TSPs with hotel selection and refer to Vansteenwegen, Souffriau, and Van Oudheusden (2011) for a profound overview on TOPs in general.

Vansteenwegen, Souffriau, and Sörensen (2012) introduced the TSP with hotel selection and developed a LS heuristic and two constructive procedures. The authors investigated a lexicographic objective function, minimizing the number of trips first and the traveled distance second. New benchmark sets were proposed and used to show the effectiveness of the presented algorithm. Li and Keskin (2014) focused on the patrol coverage for state troopers and developed an LRP that can also be handled as a TOP with hotel selection (TOPHS). The authors developed a SA heuristic and designed instances based on the crash history data in Alabama. Castro et al. (2013) developed a memetic algorithm with a TS component for the TSP with hotel selection. This algorithm strongly outperforms the LS of Vansteenwegen, Souffriau, and Sörensen (2012). Divsalar, Vansteenwegen, and Cattrysse (2013) developed a VNS with an LS component for the orienteering problem with hotel selection. The authors created a large benchmark set of 224 instances to evaluate the performance of their algorithm. Divsalar et al. (2014b) proposed a memetic algorithm which clearly outperforms all other approaches on these benchmark instances. In addition, the authors developed 176 additional large-sized instances. This algorithm was also used in Divsalar et al. (2014a) to derive personalized multi-day trips in touristic regions. Baltz et al. (2015) studied the TSP with hotel selection and multiple time windows, proposing a cheapest insertion heuristic. The authors evaluated the algorithm on existing benchmark instances and used the algorithm to investigate an additional real-world case.

#### 4.3. Analysis

In the following, we analyze all publications summarized in Section 4.1 and Section 4.2 while focusing on their most important characteristics.

Table 13 and Figure 9 highlight the scope of these publications. As can be seen, 37% of the publications focus on case studies, while 63% of the publications focus on a methodological contribution by introducing a new problem or a new algorithm for a certain problem class. All of the analyzed publications are based on a node routing problem formulation due to the addressed



Figure 9 Characteristics of publications on intermediate stops for rests and breaks.

Case study Methodological	<ul> <li>Ceselli, Righini, and Salani (2009), Kok, Hans, and Schutten (2011), Rancourt, Cordeau, and Laporte (2013), Divsalar et al. (2014a), Li and Keskin (2014), Baltz et al. (2015), Schiffer et al. (2017a)</li> <li>Xu et al. (2003), Goel (2009), Kok et al. (2010), Prescott-Gagnon et al. (2010), Vansteenwegen, Souffriau, and Sörensen (2012), Castro et al. (2013), Divsalar, Vansteenwegen, and Cattrysse (2013), Divsalar et al. (2014b), Goel and Vidal (2014), Goel and Irnich (2016), Koç, Jabali, and Laporte (2017)</li> </ul>
Node routing Arc routing	Xu et al. (2003), Ceselli, Righini, and Salani (2009), Goel (2009), Kok et al. (2010), Prescott-Gagnon et al. (2010), Kok, Hans, and Schutten (2011), Vansteenwegen, Souffriau, and Sörensen (2012), Castro et al. (2013), Divsalar, Vansteenwegen, and Cattrysse (2013), Rancourt, Cordeau, and Laporte (2013), Divsalar et al. (2014b,a), Goel and Vidal (2014), Li and Keskin (2014), Baltz et al. (2015), Goel and Irnich (2016), Koç, Jabali, and Laporte (2017), Schiffer et al. (2017a) none
HOS regulations Other	Xu et al. (2003), Ceselli, Righini, and Salani (2009), Goel (2009), Kok et al. (2010), Prescott-Gagnon et al. (2010), Kok, Hans, and Schutten (2011), Rancourt, Cordeau, and Laporte (2013), Goel and Vidal (2014), Goel and Irnich (2016), Koç, Jabali, and Laporte (2017), Schiffer et al. (2017a) Vansteenwegen, Souffriau, and Sörensen (2012), Castro et al. (2013), Divsalar, Vansteenwegen, and Cattrysse (2013), Divsalar et al. (2014b,a), Li and Keskin (2014), Baltz et al. (2015)
Deterministic Uncertain	Xu et al. (2003), Ceselli, Righini, and Salani (2009), Goel (2009), Kok et al. (2010), Prescott-Gagnon et al. (2010), Kok, Hans, and Schutten (2011), Rancourt, Cordeau, and Laporte (2013), Goel and Vidal (2014), Goel and Irnich (2016), Koç, Jabali, and Laporte (2017), Vansteenwegen, Souffriau, and Sörensen (2012), Castro et al. (2013), Divsalar, Vansteenwegen, and Cattrysse (2013), Divsalar et al. (2014b,a), Li and Keskin (2014), Baltz et al. (2015), Schiffer et al. (2017a) none
VRP TSP / TOP	Xu et al. (2003), Ceselli, Righini, and Salani (2009), Goel (2009), Kok et al. (2010), Prescott-Gagnon et al. (2010), Kok, Hans, and Schutten (2011), Rancourt, Cordeau, and Laporte (2013), Goel and Vidal (2014), Goel and Irnich (2016), Koç, Jabali, and Laporte (2017), Schiffer et al. (2017a) Vansteenwegen, Souffriau, and Sörensen (2012), Castro et al. (2013), Divsalar, Vansteenwegen, and Cattrysse (2013), Divsalar et al. (2014b,a), Li and Keskin (2014), Baltz et al. (2015)

 Table 13
 Scope of publications on intermediate stops for rests and breaks.

application areas (logistics fleets and trips with special points of interest). None of the analyzed publications considers uncertain data. All papers that consider HOS regulations are based on a VRP approach (63%) because logistics fleets are analyzed (cf. Figure 9). All approaches that focus on other application cases than logistics fleets are modeled as a TSP or TOP.

Figure 10 and Table 14 detail the shares of the different problem variants arising in the context of intermediate stops for idling. As time is the operational resource related to those stops, all problem variants assume arc-based consumption. Most papers consider a maximum route duration (42%) or time windows (42%), which arise out of a long-haul logistics context or a maximum trip duration for e.g., tourist trips. The problem variant ADPIFTW belongs to the EVRPTDS as discussed in Schiffer et al. (2017a) which is the only variant that addresses a goal conflict between two operational resources (driver time and energy). Figure 11 and Table 15 detail the objectives. While approximately 50% of the publications focus on a cost or duration minimization, there are also many papers that use a lexicographic objective, minimizing the number of vehicles first and the total distance second. These publications often focus on the VRTDSP. Only the TOP variants maximize the score of a trip.



Figure 10 Types of problem variants for intermediate stops for rests and breaks.



AIF	Xu et al. (2003)
AIFMRD	Ceselli, Righini, and Salani (2009), Vansteenwegen, Souffriau, and Sörensen (2012), Castro et al. (2013), Divsalar,
	Vansteenwegen, and Cattrysse (2013), Divsalar et al. (2014b,a), Baltz et al. (2015), Koç, Jabali, and Laporte (2017)
AIFTW	Goel (2009), Kok et al. (2010), Prescott-Gagnon et al. (2010), Kok, Hans, and Schutten (2011), Rancourt, Cordeau, and
	Laporte (2013), Goel and Vidal (2014), Li and Keskin (2014), Goel and Irnich (2016)
ADPIFTW	Schiffer et al. (2017a)



Figure 11 Objectives for intermediate stops for break and rest periods.

Table 15	Objectives for	intermediate sto	ns for break an	d rest neriods
Table 15	Objectives for	interneulate sto	ps for break an	u rest perious.

Objective	References
Costs	Xu et al. (2003), Ceselli, Righini, and Salani (2009), Baltz et al. (2015), Goel and Irnich (2016), Schiffer et al. (2017a)
Distance	Goel (2009), Kok et al. (2010), Prescott-Gagnon et al. (2010), Vansteenwegen, Souffriau, and Sörensen (2012), Castro et al. (2013), Goel and Vidal (2014)
Duration	Kok, Hans, and Schutten (2011), Rancourt, Cordeau, and Laporte (2013), Li and Keskin (2014)
Num. veh. / trips	Goel (2009), Kok et al. (2010), Prescott-Gagnon et al. (2010), Vansteenwegen, Souffriau, and Sörensen (2012), Castro et al. (2013), Rancourt, Cordeau, and Laporte (2013), Goel and Vidal (2014)
Maximum scores.	Divsalar, Vansteenwegen, and Cattrysse (2013), Divsalar et al. (2014b,a)

The solution methods listed in Table 16 cover a wide variety of exact algorithms and metaheuristics, including matheuristics. Some publications propose MIPs and most exact solution approaches are based on BP algorithms. Also, matheuristics combining CG with a powerful metaheuristic component are often used and provide state-of-the-art results (cf. Prescott-Gagnon et al. 2010). Among metaheuristics, genetic algorithms and ALNS or VNS based algorithms yield the best results. LS algorithms are successfully used for TOP and TSP variants (cf. Vansteenwegen, Souffriau, and Sörensen 2012, Divsalar, Vansteenwegen, and Cattrysse 2013).

The benchmark instance sets are summarized in Table 17 for the VRP variants and in Table 18 for the TSP and TOP variants. The tables show the number of instances, the number of customers, the number of intermediate stop options (if applicable) and other papers that use the same instances. For the VRP variants, besides some early instance sets of Xu et al. (2003) and Ceselli, Righini, and Salani (2009), most publications use the instance set provided in Goel (2009). Only Schiffer et al. (2017a) propose different instance sets, limited to a one-day planning horizon in mid-haul logistics and accounting for the characteristics of ECVs. Instance sets for the TSP variants with up to 1002 customers are given in Vansteenwegen, Souffriau, and Sörensen (2012), and the most common benchmarks for the TOP with hotel selection are presented in Divsalar, Vansteenwegen, and Cattrysse (2013).

Table 16 Solution methods for intermediate stops for break and rest periods.

Exact	
(M)I(L)P	Kok, Hans, and Schutten (2011), Vansteenwegen, Souffriau, and Sörensen (2012), Divsalar, Vansteenwegen, and Cattrysse (2013), Li and Keskin (2014)
BP / CG	Xu et al. (2003), Ceselli, Righini, and Salani (2009), Prescott-Gagnon et al. (2010), Goel and Irnich (2016)
lower bound	Xu et al. (2003)
DP	Xu et al. (2003), Ceselli, Righini, and Salani (2009), Kok et al. (2010), Goel and Irnich (2016)
	Heuristic
TS	Prescott-Gagnon et al. (2010), Castro et al. (2013), Rancourt, Cordeau, and Laporte (2013)
(A)LNS	Goel (2009), Prescott-Gagnon et al. (2010), Schiffer et al. (2017a)
(A)VNS	Divsalar, Vansteenwegen, and Cattrysse (2013), Divsalar et al. (2014a)
LS	Vansteenwegen, Souffriau, and Sörensen (2012), Divsalar, Vansteenwegen, and Cattrysse (2013), Schiffer et al. (2017a)
$\mathbf{EA}$	Castro et al. (2013), Divsalar et al. (2014b,a), Goel and Vidal (2014)
SA	Li and Keskin (2014)
Other	Xu et al. (2003), Kok et al. (2010), Kok, Hans, and Schutten (2011), Baltz et al. (2015)

Abbreviations hold as follows: (M)I(L)P - (mixed) integer (linear) program; DP - dynamic programing; BP - branch and price; CG - column generation; LB - lower bound, TS - tabu search; (A)LNS - (adaptive) large neighborhood search; (A)VNS - (adaptive) variable neighborhood search; LS - local search; EA - evolutionary algorithm; SA - Simulated Annealing.

		Tuble II	msta		
Reference	Type	Ι	С	IF	Used within
Xu et al. (2003)	AIF	19	50-210		
Xu et al. (2003)	AIF	15	300-500		
Ceselli, Righini, and Salani (2009)	AIFMRD	46	1-47		
Goel(2009)	AIFTW	56	100		Kok et al. (2010), Prescott-Gagnon et al. (2010), Kok, Hans, and Schutten (2011), Goel and Vidal (2014), Goel and Irnich (2016)
Rancourt, Cordeau, and Laporte (2013)	AIFTW	1	162		
Rancourt, Cordeau, and Laporte (2013)	AIFTW	56	100		
Schiffer et al. (2017a)	ADPIFTW	56	100	897	

Table 17 Instances for the VRTDSP.

Reference	Type	Ι	С	IF	Used within
Vansteenwegen, Souffriau, and Sörensen (2012)	AIFMRD	16	48 - 288	5	Castro et al. (2013), Baltz et al. (2015)
Vansteenwegen, Souffriau, and Sörensen (2012)	AIFMRD	16	10 - 40	1	Castro et al. $(2013)$ , Baltz et al. $(2015)$
Vansteenwegen, Souffriau, and Sörensen (2012)	AIFMRD	48	51 - 1002	3-10	Castro et al. $(2013)$ , Baltz et al. $(2015)$
Li and Keskin (2014)	AIFTW	32	16 - 32	4 - 32	
Divsalar, Vansteenwegen, and Cattrysse (2013)	AIFMRD	105	32 - 102	1-3	Divsalar et al. (2014b,a)
Divsalar, Vansteenwegen, and Cattrysse (2013)	AIFMRD	70	32-102	5-6	Divsalar et al. (2014b,a)
Divsalar, Vansteenwegen, and Cattrysse (2013)	AIFMRD	44	64 - 100	10-12	Divsalar et al. (2014b,a)
Divsalar et al. (2014b) Baltz et al. (2015)	AIFMRD AIFMRD	$\begin{array}{c} 176 \\ 210 \end{array}$	${}^{65-130}_{5-50}$	$_{5-50}^{3-15}$	Divsalar et al. (2014a)

Table 18 Instances for TSPs / TOPs with hotel selection.

Concluding, problems arising in the context of routing with intermediate stops for idling, namely the VRTDSP as well as TSPs and TOPs with hotel selection, remain a young and active research field which has gained interest in recent years. Methodologically, problem variants stick to the already addressed ones (mainly AIFMRD and AIFTW) due to the application cases. However, more complex variants like the ADPIFTW resulting out of the EVRPTDS may arise when HOS regulations or other idling options are integrated into rich routing problems. Generally, specific case studies on the (known) methodologically proposed problem variants are still sparse. For the VRTDSP, it is mainly long-haul routing that has been addressed until now (except for Schiffer et al. (2017a)). Aiming at real-world cases in rich VRPs, integrating HOS regulations into midhaul routing seems to be an interesting option. Furthermore, analyzing more specific regulations (e.g., as in Kok et al. (2010)) may inspire researchers in this field. Besides these aspects, considering autonomous driving assistance may lead to interesting future research questions. Although completely autonomously driving cars make the consideration of HOS regulations obsolete, any intermediate level of autonomous driving offers interesting synchronization problems between (temporally or spatially limited) autonomous driving periods during which the driver can rest and conventional driving periods.

#### 5. Conclusion

We have studied RPISs. An intermediate stop has been defined to be a stop en route that is not related to providing service, but is necessary to keep the vehicle operational. This definition was used to distinguish this problem class from multi-echelon problem variants. Routing problems for intermediate stops have been divided into three large application areas, namely i) replenishment and unloading, ii) refueling, and iii) idling for breaks and rests. A nomenclature to characterize routing problems with intermediate stops has been introduced and used to point out the decisive modeling characteristics of these problems. Furthermore, algorithmic findings have been identified for the different problem variants, and effective solution techniques have been highlighted, e.g., adding dynamic programming components to optimally locate intermediate stops on routes in metaheuristics.

This survey showed that the three application areas have known different temporal developments. While RPIS for replenishment and unloading have constantly been investigated due to the relevance in municipal services and classical logistics services, RPIS for refueling and idling have been ignored for a long time. Routing problems for refueling have traditionally been limited to ICEVs, but have become popular in recent years due to the increasing role of AFVs, especially ECVs, in sustainable transportation. RPIS for idling have known a peak in recent years as researchers have paid increasing attention to rich VRPs.

Besides providing an in-depth overview on routing problems with intermediate stops, this survey has identified some promising directions for future research. Briefly summarized, these directions are as follows.

For intermediate stops for replenishment or unloading, online algorithms for same-day and express deliveries constitute an interesting field of research. Additionally, the consideration of uncertainties, especially for customer patterns, seems to be a promising research direction for both classical replenishment networks as well as city logistics applications.

For intermediate stops for refueling, again, including uncertainty into the addressed problem classes seems to be a challenging and interesting research direction. Only a few recent publications include uncertain customer patterns (cf. Schiffer and Walther 2017c) or uncertain charging behavior (cf. Sweda, Dolinskaya, and Klabjan 2017). Uncertain travel times have not been considered so far but may be crucial to derive feasible route plans for ECVs. Moreover, real-world case studies on the usage of ECVs in logistics fleets are still sparse. Future works on case studies may pave the way for a market uptake of ECVs and thus, more sustainable and environmentally friendly means of transportation.

For intermediate stops for rests and breaks, future research may aim at more elaborate case studies. Additionally, the consideration of different levels of autonomous driving vehicles constitutes a promising direction for future research.

While these research directions are tailored to one application area, some research directions that combine the different application areas in order to realize synchronization potentials result. The first LRP approaches that highlight the interdependencies between charging station location and vehicle routing decisions have just been published. However, the resulting improvement potential has so far been neglected in network design for intermediate stops for replenishing and unloading. Regarding this issue may reveal significant cost improvements for these application cases. The first LRP paper that highlights such improvements has recently been published by Schiffer, Schneider, and Laporte (2017). Focusing on ECVs, a first paper on the synchronization potential between ECV charging and breaks resulting from HOS regulations has been published by Schiffer et al. (2017a).

# Acknowledgments

Gilbert Laporte was partially funded by the Canadian Natural Sciences and Engineering Research Council under grant 2015-06189. This support is gratefully acknowledged. Thanks are due to two anonymous referees for their valuable comments.

# Appendix

#### Table 19 Acronyms.

AFV	alternative fuel vehicle
ALNS	adaptive LNS
BC	branch-and-cut
BP	branch-and-price
BPC	branch-and-price-and-cut
BSS-EV-LRP	battery swap station electric vehicle LRP
CARP	capacitated arc routing problem
CARPIF	CARP with intermediate facilities
CARPRP	CARP with refill points
CG	column generation
$\mathrm{DP}$	dynamic programming
$\mathrm{ECV}$	electric commercial vehicle
ELRPTWPR	electric LRP with time windows and partial recharging
EU	European Union
EVRP	electric VRP
EVRPTDS	EVRP with truck driver scheduling
EVRPTW	EVRP with time windows
EVRPTWPR	EVRPTW and partial recharging
GRASP	greedy randomized adaptive search procedure
GVRP	green VRP
HEV	hybrid electric vehicle
HOS	hours of service
ICEV	internal combustion engine vehicle
LNS	large neighborhood search
LRP	location-routing problem
LRPIF	LRP with intra-route facilities
LS	local search
MDVRPI	multi-depot VRP with inter-depot routes
MIP	mixed integer program
PVRP	periodic VRP
RPIS	routing problems with intermediate stops
SA	simulated annealing
SPP	shortest path problem
TOP	team orienteering problem
TS	tabu search
TSP	traveling salesman problem
TTRP	truck and trailer routing problem
US	United States
VNS	variable neighborhood search
VRP	vehicle routing problem
VRPIS	VRP with intermediate stops
VRPTW	VRP with time windows
VRTDSP	vehicle routing and truck driver scheduling problem

#### References

- Albareda-Sambola M, 2015 Location-routing and location-arc routing. Laporte G, Nickel S, Saldanha da Gama F, Albareda-Sambola M, eds., Location Science, 399–418 (Berlin: Springer).
- Amaya CA, Langevin A, Trépanier M, 2007 The capacitated arc routing problem with refill points. Operations Research Letters 35(1):45–53.
- Amaya CA, Langevin A, Trépanier M, 2010 A heuristic method for the capacitated arc routing problem with refill points and multiple loads. Journal of the Operational Research Society 61(7):1095–1103.
- Andelmin J, Bartolini, 2017 An exact algorithm for the green vehicle routing problem. Transportation Science Articles in Advance.
- Angelelli E, Speranza MG, 2002a The application of a vehicle routing model to a waste-collection problem: two case studies. Journal of the Operational Research Society 53(9):944–952.
- Angelelli E, Speranza MG, 2002b The periodic vehicle routing problem with intermediate facilities. European Journal of Operational Research 137(2):233–247.
- Archetti C, Speranza MG, Vigo D, 2014 Vehicle routing problems with profits. Toth P, Vigo D, eds., Vehicle Routing: Problems, Methods, and Applications, 273–297 (SIAM).
- Arslan O, Yildiz B, Karasan OE, 2015 Minimum cost path problem for plug-in hybrid electric vehicles. Transportation Research Part E: Logistics and Transportation Review 80:123–141.
- Baltz A, El Ouali M, Jäger G, Sauerland V, Srivastav A, 2015 Exact and heuristic algorithms for the travelling salesman problem with multiple time windows and hotel selection. Journal of the Operational Research Society 66(4):615–626.
- Barco J, Guerra A, Muñoz L, Quijano N, 2013 Optimal routing and scheduling of charge for electric vehicle: Case study. arXiv preprint arXiv:1310.0145.
- Bard J, Huang L, Dror M, Jaillet P, 1998 A branch and cut algorithm for the VRP with satellite facilities. IIE Transactions 30(9):821–834.
- Bartolini E, Schneider M, 2017 A two-commodity flow formulation for the capacitated truck-and-trailer routing problem. Working Paper DPO-2017-02, RWTH Aachen.
- Belenguer JM, Benavent E, Martínez A, Prins C, Prodhon C, Villegas JG, 2016 A branch-and-cut algorithm for the single truck and trailer routing problem with satellite depots. Transportation Science 50(2):735–749.
- Beliën J, De Boeck L, Van Ackere J, 2014 Municipal solid waste collection and management problems: A literature review. Transportation Science 48(1):78–102.
- Beltrami EJ, Bodin LD, 1974 Networks and vehicle routing for municipal waste collection. Networks 4(1):65-94.
- Benjamin AM, Beasley JE, 2010 Metaheuristics for the waste collection vehicle routing problem with time windows, driver rest period and multiple disposal facilities. Computers & Operations Research 37(12):2270–2280.
- Bousonville T, Hartmann A, Melo T, Kopfer H, 2011 Vehicle routing and refueling: The impact of price variations on tour length. Logistikmanagement - Herausforderungen, Chancen, und Lösungen (Proceedings of LM 11), volume 2, 83–101.
- Bruglieri M, Pezzella F, Pisacane O, Suraci S, 2015a A matheuristic for the electric vehicle routing problem with time windows. arXiv preprint arXiv:1506.00211.
- Bruglieri M, Pezzella F, Pisacane O, Suraci S, 2015b A variable neighborhood search branching for the electric vehicle routing problem with time windows. Electronic Notes in Discrete Mathematics 47:221–228.
- Buhrkal K, Larsen A, Ropke S, 2012 The waste collection vehicle routing problem with time windows in a city logistics context. Procedia - Social and Behavioral Sciences, Seventh International Conference on City Logistics, June 7- 9,2011, Mallorca, Spain, volume 39, 241–254.
- Caramia M, Guerriero F, 2010 A heuristic approach for the truck and trailer routing problem. Journal of the Operational Research Society 61(7):1168–1180.
- Castro M, Sörensen K, Vansteenwegen P, Goos P, 2013 A memetic algorithm for the travelling salesperson problem with hotel selection. Computers & Operations Research 40(7):1716–1728.
- Ceselli A, Righini G, Salani M, 2009 A column generation algorithm for a rich vehicle-routing problem. Transportation Science 43(1):56–69.
- Chao IM, 2002 A tabu search method for the truck and trailer routing problem. Computers & Operations Research 29(1):33–51.

- Clarke G, Wright JW, 1964 Scheduling of vehicles from a central depot to a number of delivery points. Operations Research 12(4):568–581.
- Coene S, Arnout A, Spieksma FCR, 2010 On a periodic vehicle routing problem. Journal of the Operational Research Society 61(12):1719–1728.
- Conrad RG, Figliozzi MA, 2011 The recharging vehicle routing problem. Doolen T, Van Aken E, eds., Proceedings of the 2011 Industrial Engineering Research Conference, 1–8 (Reno, NV).
- Contardo C, Hemmelmayr V, Crainic TG, 2012 Lower and upper bounds for the two-echelon capacitated locationrouting problem. Computers & Operations Research 39(12):3185–3199.
- Crainic TG, Ricciardi N, Storchi G, 2009 Models for evaluating and planning city logistics systems. Transportation Science 43(4):432–454.
- Crevier B, Cordeau JF, Laporte G, 2007 The multi-depot vehicle routing problem with inter-depot routes. European Journal of Operational Research 176(2):756–773.
- Cuda R, Guastaroba G, Speranza MG, 2015 A survey on two-echelon routing problems. Computers & Operations Research 55:185–199.
- Daskin M, 1995 Network and Discrete Location: Models, Algorithms, and Applications (New York: Wiley).
- De Rosa B, Improta G, Ghiani G, Musmanno R, 2002 The arc routing and scheduling problem with transshipment. Transportation Science 36(3):301–313.
- Del Pia A, Filippi C, 2006 A variable neighborhood descent algorithm for a real waste collection problem with mobile depots. International Transactions in Operational Research 13(2):125–141.
- Derigs U, Pullmann M, Vogel U, 2013 Truck and trailer routing problems, heuristics and computational experience. Computers & Operations Research 40(2):536–546.
- Desaulniers G, Desrosiers J, Solomon MM, eds., 2005 Column Generation (New York: Springer).
- Desaulniers G, Errico F, Irnich S, Schneider M, 2016 Exact algorithms for electric vehicle-routing problems with time windows. Operations Research 64(6):1388–1405.
- Divsalar A, Vansteenwegen P, Cattrysse D, 2013 A variable neighborhood search method for the orienteering problem with hotel selection. International Journal of Production Economics 145(1):150–160.
- Divsalar A, Vansteenwegen P, Chitsaz M, Sörensen K, Cattrysse D, 2014a Personalized multi-day trips to touristic regions: A hybrid GA-VND approach. Blum C, Ochoa G, eds., Evolutionary Computation in Combinatorial Optimisation: 14th European Conference, EvoCOP 2014, Granada, Spain, April 23-25, 2014, Revised Selected Papers, 194–205 (Berlin, Heidelberg: Springer), ISBN 978-3-662-44320-0.
- Divsalar A, Vansteenwegen P, Sörensen K, Cattrysse D, 2014b A memetic algorithm for the orienteering problem with hotel selection. European Journal of Operational Research 237(1):29–49.
- Dohn A, Kolind E, Clausen J, 2009 The manpower allocation problem with time windows and job-teaming constraints: A branch-and-price approach. Computers & Operations Research 36(4):1145–1157.
- Doppstadt C, Koberstein A, Vigo D, 2016 The hybrid electric vehicle traveling salesman problem. European Journal of Operational Research 253(3):825–842.
- Drexl M, 2012 Synchronization in vehicle routing—A survey of VRPs with multiple synchronization constraints. Transportation Science 46(3):297–316.
- Drexl M, 2013 Applications of the vehicle routing problem with trailers and transshipments. European Journal of Operational Research 227(2):275–283.
- Drexl M, 2014 Branch-and-cut algorithms for the vehicle routing problem with trailers and transhipments. Networks 63(1):119–133.
- Drexl M, Schneider M, 2014 A survey of variants and extensions of the location-routing problem. European Journal of Operational Research 241(2):283–308.
- Erdoğan S, Miller-Hooks E, 2012 A green vehicle routing problem. Transportation Research Part E: Logistics and Transportation Review 48(1):100–114.
- Felipe A, Ortuño MT, Righini G, Tirado G, 2014 A heuristic approach for the green vehicle routing problem with multiple technologies and partial recharges. Transportation Research Part E: Logistics and Transportation Review 71:111–128.
- Froger A, Mendoza JE, Jabali O, Laporte G, 2017a New formulations for the electric vehicle routing problem with nonlinear charging functions. CIRRELT-2017-30.

- Froger A, Mendoza JE, Jabali O, Laporte G, 2017b A matheuristic for the electric vehicle routing problem with capacitated charging stations. CIRRELT-2017-31.
- Fügenschuh A, 2009 Solving a school bus scheduling problem with integer programming. European Journal of Operational Research 193(3):867–884.
- Funke B, Grünert T, Irnich S, 2005 Local search for vehicle routing and scheduling problems: Review and conceptual integration. Journal of Heuristics 11(4):267–306.
- Gendreau M, Potvin JY, 2010 Handbook of Metaheuristics, volume 2 of International Series in Operations Research and Management Science (New York: Springer).
- Gerdessen JC, 1996 Vehicle routing problem with trailers. European Journal of Operational Research 93(1):135–147.
- Ghiani G, Guerriero F, Improta G, Musmanno R, 2005 Waste collection in southern Italy: Solution of a real-life arc routing problem. International Transactions in Operational Research 12(2):135–144.
- Ghiani G, Guerriero F, Laporte G, Musmanno R, 2004 Tabu search heuristics for the arc routing problem with intermediate facilities under capacity and length restrictions. Journal of Mathematical Modelling and Algorithms 3(3):209–223.
- Ghiani G, Improta G, Laporte G, 2001 The capacitated arc routing problem with intermediate facilities. Networks 37(3):134–143.
- Ghiani G, Laganà D, Laporte G, Mari F, 2010 Ant colony optimization for the arc routing problem with intermediate facilities under capacity and length restrictions. Journal of Heuristics 16(2):211–233.
- Goeke D, Schneider M, 2015 Routing a mixed fleet of electric and conventional vehicles. European Journal of Operational Research 245(1):81–99.
- Goel A, 2009 Vehicle scheduling and routing with drivers' working hours. Transportation Science 43(1):17–26.
- Goel A, Irnich S, 2016 An exact method for vehicle routing and truck driver scheduling problems. Transportation Science.
- Goel A, Vidal T, 2014 Hours of service regulations in road freight transport: An optimization-based international assessment. Transportation Science 48(3):391–412.
- Golden BL, Raghavan S, Wasil EA, eds., 2008 The Vehicle Routing Problem: Latest Advances and New Challenges, volume 43 of Operations Research/Computer Science Interfaces (Berlin: Springer).
- Gonçalves F, Cardoso SR, Relvas S, Barbosa-Póvoa APFD, 2011 Optimization of a distribution network using electric vehicles: A VRP problem. IO2011 - 15° Congresso da associação Portuguesa de Investigação Operacional, 18–20.
- Gørtz IL, Nagarajan V, Ravi R, 2009 Minimum makespan multi-vehicle dial-a-ride. Fiat A, Sanders P, eds., Algorithms
   ESA 2009: 17th Annual European Symposium, Copenhagen, Denmark, September 7-9, 2009. Proceedings, 540–552 (Berlin, Heidelberg: Springer), ISBN 978-3-642-04128-0.
- Guastaroba G, Speranza MG, Vigo D, 2016 Intermediate facilities in freight transportation planning: A survey. Transportation Science 50(3):763–789.
- Hemmelmayr V, Doerner KF, Hartl RF, Rath S, 2013a A heuristic solution method for node routing based solid waste collection problems. Journal of Heuristics 19(2):129–156.
- Hemmelmayr VC, Doerner KF, Hartl RF, Vigo D, 2013b Models and algorithms for the integrated planning of bin allocation and vehicle routing in solid waste management. Transportation Science 48(1):103–120.
- Hiermann G, Puchinger J, Ropke S, Hartl RF, 2016 The electric fleet size and mix vehicle routing problem with time windows and recharging stations. European Journal of Operational Research 252(3):995–1018.
- Hof J, Schneider M, Goeke D, 2017 Solving the battery swap station location-routing problem with capacitated electric vehicles using an AVNS algorithm for vehicle-routing problems with intermediate stops. Transportation Research Part B: Methodological 97:102–112.
- Ichimori T, Ishii H, 1981 Routing a vehicle with the limitation of fuel. Journal of the Operations Research Society of Japan 24(3):277–281.
- Ichimori T, Ishii H, Nishida T, 1983 Two routing problems with the limitation of fuel. Discrete Applied Mathematics 6(1):85–89.
- Jordan WC, 1987 Truck backhauling on networks with many terminals. Transportation Research Part B: Methodological 21(3):183 – 193.
- Kek AGH, Cheu RL, Meng Q, 2008 Distance-constrained capacitated vehicle routing problems with flexible assignment of start and end depots. Mathematical and Computer Modelling 47(1-2):140-152.

- Keskin M, Çatay B, 2016 Partial recharge strategies for the electric vehicle routing problem with time windows. Transportation Research Part C: Emerging Technologies 65:111–127.
- Khuller S, Malekian A, Mestre J, 2011 To fill or not to fill: The gas station problem. ACM Transactions on Algorithms 7(3):1–16.
- Kim BI, Kim S, Sahoo S, 2006 Waste collection vehicle routing problem with time windows. Computers & Operations Research 33(12):3624–3642.
- Koç Ç, Jabali O, Laporte G, 2017 Long-haul vehicle routing and scheduling with idling options. Journal of the Operational Research Society :1–13.
- Koç Ç, Karaoglan I, 2016 The green vehicle routing problem: A heuristic based exact solution approach. Applied Soft Computing 39:154–164.
- Kok AL, Hans EW, Schutten JMJ, 2011 Optimizing departure times in vehicle routes. European Journal of Operational Research 210(3):579–587.
- Kok AL, Meyer CM, Kopfer H, Schutten JMJ, 2010 A dynamic programming heuristic for the vehicle routing problem with time windows and European Community social legislation. Transportation Science 44(4):442–454.
- Laporte G, Martello S, 1990 The selective traveling salesman problem. Discrete Applied Mathematics 26(2):193–207.
- Laporte G, Nickel S, Saldanha da Gama F, Albareda-Sambola M, eds., 2015 Location Science (Berlin: Springer).
- Li S, Keskin B, 2014 Bi-criteria dynamic location-routing problem for patrol coverage. Journal of the Operational Research Society 65(11):1711–1725.
- Liao C, Lu SH, Shen ZJM, 2016 The electric vehicle touring problem. Transportation Research Part B: Methodological 86:163–180.
- Lin J, Zhou W, Wolfson O, 2016 Electric vehicle routing problem. Transportation Research Procedia 12:508–521.
- Lopes RB, Ferreira C, Santos BS, Barreto S, 2013 A taxonomical analysis, current methods and objectives on locationrouting problems. International Transactions in Operational Research 20(6):795–822.
- Mancini S, 2017 The hybrid vehicle routing problem. Transportation Research Part C: Emerging Technologies 78:1–12.
- Markov I, Varone S, Bierlaire M, 2016 Integrating a heterogeneous fixed fleet and a flexible assignment of destination depots in the waste collection VRP with intermediate facilities. Transportation Research Part B: Methodological 84:256–273.
- Montoya A, Guéret C, Mendoza JE, Villegas JG, 2016 A multi-space sampling heuristic for the green vehicle routing problem. Transportation Research Part C: Emerging Technologies 70:113–128.
- Montoya A, Guéret C, Mendoza JE, Villegas JG, 2017 The electric vehicle routing problem with nonlinear charging function. Transportation Research Part B: Methodological, in Press.
- Mourão MC, Almeida MT, 2000 Lower-bounding and heuristic methods for a refuse collection vehicle routing problem. European Journal of Operational Research 121(2):420–434.
- Mourão MC, Amado L, 2005 Heuristic method for a mixed capacitated arc routing problem: A refuse collection application. European Journal of Operational Research 160(1):139–153.
- Muter I, Cordeau JF, Laporte G, 2014 A branch-and-price algorithm for the multidepot vehicle routing problem with interdepot routes. Transportation Science 48(3):425–441.
- Nejad MM, Mashayekhy L, Grosu D, Babu Chinnam R, 2017 Optimal routing for plug-in hybrid electric vehicles. Transportation Science, articles in Advance.
- Parragh SN, Cordeau JF, 2017 Branch-and-price and adaptive large neighborhood search for the truck and trailer routing problem with time windows. Computers & Operations Research 83:28–44.
- Pelletier S, Jabali O, Laporte G, Veneroni M, 2017 Battery degradation and behaviour for electric vehicles: Review and numerical analyses of several models. Transportation Research Part B: Methodological, in Press.
- Perrier N, Langevin A, Campbell JF, 2006 A survey of models and algorithms for winter road maintenance. Part I: System design for spreading and plowing. Computers & Operations Research 33(1):209–238.
- Perrier N, Langevin A, Campbell JF, 2007 A survey of models and algorithms for winter road maintenance. Part IV: Vehicle routing and fleet sizing for plowing and snow disposal. Computers & Operations Research 34(1):258–294.
- Polacek M, Doerner KF, Hartl RF, Maniezzo V, 2008 A variable neighborhood search for the capacitated arc routing problem with intermediate facilities. Journal of Heuristics 14:405–423.
- Prescott-Gagnon E, Desaulniers G, Drexl M, Rousseau LM, 2010 European driver rules in vehicle routing with time windows. Transportation Science 44(4):455–473.

- Prescott-Gagnon E, Desaulniers G, Rousseau LM, 2014 Heuristics for an oil delivery vehicle routing problem. Flexible Services and Manufacturing Journal 26(4):516–539.
- Prodhon C, Prins C, 2014 A survey of recent research on location-routing problems. European Journal of Operational Research 238(1):1–17.
- Rais A, Alvelos F, Carvalho MS, 2014 New mixed integer-programming model for the pickup-and-delivery problem with transhipment. European Journal of Operational Research 235(3):530–539.
- Rancourt ME, Cordeau JF, Laporte G, 2013 Long-haul vehicle routing and scheduling with working hour rules. Transportation Science 47(1):81–107.
- Roberti R, Wen M, 2016 The electric traveling salesman problem with time windows. Transportation Research Part E: Logistics and Transportation Review 89:32–52.
- Rothenbächer AK, Drexl M, Irnich S, 2016 Branch-and-price-and-cut for the truck-and-trailer routing problem with time windows. Transportation Science.
- Salazar-Aguilar MA, Langevin A, Laporte G, 2012 Synchronized arc routing for snow plowing operations. Computers & Operations Research 39(7):1432–1440.
- Salazar-Aguilar MA, Langevin A, Laporte G, 2013 The synchronized arc and node routing problem: Application to road marking. Computers & Operations Research 40(7):1708–1715.
- Santos L, Coutinho-Rodrigues J, Current JR, 2008 Implementing a multi-vehicle multi-route spatial decision support system for efficient trash collection in Portugal. Transportation Research Part A: Policy and Practice 42(6):922– 934.
- Sassi O, Cherif-Khettaf WR, Oulamara A, 2015a Iterated tabu search for the mix fleet vehicle routing problem with heterogenous electric vehicles. Le Thi HA, Pham Dinh T, Nguyen NT, eds., Modelling, Computation and Optimization in Information Systems and Management Sciences, volume 359 of Advances in Intelligent Systems and Computing, 57–68 (Springer International Publishing).
- Sassi O, Cherif-Khettaf WR, Oulamara A, 2015b Multi-start iterated local search for the mixed fleet vehicle routing problem with heterogenous electric vehicles. Ochoa G, Chicano F, eds., Evolutionary Computation in Combinatorial Optimization: 15th European Conference, EvoCOP 2015, Copenhagen, Denmark, April 8–10, 2015, Proceedings, 138–149 (Springer International Publishing).
- Sassi O, Cherif-Khettaf WR, Oulamara A, 2015c Vehicle routing problem with mixed fleet of conventional and heterogenous electric vehicles and time dependent charging costs. International Journal of Mathematical, Computational, Physical, Electrical and Computer Engineering 9(3):18–28.
- Scheuerer S, 2006 A tabu search heuristic for the truck and trailer routing problem. Computers & Operations Research 33(4):894–909.
- Schiffer M, Laporte G, Schneider M, Walther G, 2017a The impact of synchronizing driver breaks and recharging operations for electric vehicles. Les Cahiers du GERAD G-2017-46.
- Schiffer M, Klein P, Schneider M, Walther G, 2017b A solution framework for a class of vehicle routing problems with intermediate stops. Working Paper OM-DPO 01/2017, RWTH Aachen.
- Schiffer M, Schneider M, Laporte G, 2017 Designing sustainable mid-haul logistics networks with intra-route multiresource facilities. Les Cahiers du GERAD G-2017-21.
- Schiffer M, Stütz S, Walther G, 2016 Are ECVs breaking even? Competitiveness of electric commercial vehicles in medium-duty logistics networks. Working Paper OM-02/2016, RWTH Aachen.
- Schiffer M, Walther G, 2017a An adaptive large neighborhood search for the location-routing problem with intra-route facilities. Transportation Science, articles in advance.
- Schiffer M, Walther G, 2017b The electric location routing problem with time windows and partial recharging. European Journal of Operational Research 260(3):995–1013.
- Schiffer M, Walther G, 2017c Strategic planning of electric logistic networks: A robust location routing approach. Omega.
- Schneider M, Drexl M, 2017 A survey of the standard location-routing problem. Annals of Operations Research 1-26.
- Schneider M, Stenger A, Goeke D, 2014 The electric vehicle-routing problem with time windows and recharging stations. Transportation Science 48(4):500–520.
- Schneider M, Stenger A, Hof J, 2015 An adaptive VNS algorithm for vehicle routing problems with intermediate stops. OR Spectrum 37(2):353–387.
- Semet F, Taillard E, 1993 Solving real-life vehicle routing problems efficiently using tabu search. Annals of Operations Research 41(4):469–488.

- Solomon MM, 1987 Algorithms for the vehicle routing and scheduling problems with time window constraints. Operations Research 35(2):254–265.
- Suzuki Y, 2012 A decision support system of vehicle routing and refueling for motor carriers with time-sensitive demands. Decision Support Systems 54(1):758–767.
- Suzuki Y, Dai J, 2012 Reducing the fuel cost of motor carriers by using optimal routing and refueling policies. Transportation Journal 51:145–163.
- Sweda TM, Dolinskaya IS, Klabjan D, 2017 Optimal recharging policies for electric vehicles. Transportation Science 51(2):457–479.
- Tan KC, Chew YH, Lee LH, 2006 A hybrid multi-objective evolutionary algorithm for solving truck and trailer vehicle routing problems. European Journal of Operational Research 172(3):855–885.
- Tarantilis CD, Zachariadis EE, Kiranoudis CT, 2008 A hybrid guided local search for the vehicle-routing problem with intermediate replenishment facilities. INFORMS Journal on Computing 20(1):154–168.
- Toth P, Vigo D, eds., 2014 Vehicle Routing: Problems, Methods, and Applications. MOS-SIAM Series on Optimization (Philadelphia, PA: SIAM), 2nd edition.
- Vansteenwegen P, Souffriau W, Sörensen K, 2012 The travelling salesperson problem with hotel selection. Journal of the Operational Research Society 63(2):207–217.
- Vansteenwegen P, Souffriau W, Van Oudheusden D, 2011 The orienteering problem: A survey. European Journal of Operational Research 209(1):1–10.
- Verma A, Lamsal K, Keough S, 2015 Electric vehicle routing problem with time windows, recharging stations and battery swapping stations. Technical report, Department of Commercial Aviation, Delta State University, Cleveland, MS, United States.
- Vidal T, Crainic TG, Gendreau M, Lahrichi N, Rei W, 2012 A hybrid genetic algorithm for multidepot and periodic vehicle routing problems. Operations Research 60(3):611–624.
- Vidal T, Crainic TG, Gendreau M, Prins C, 2013 A hybrid genetic algorithm with adaptive diversity management for a large class of vehicle routing problems with time-windows. Computers & Operations Research 40(1):475–489.
- Vidal T, Crainic TG, Gendreau M, Prins C, 2014 A unified solution framework for multi-attribute vehicle routing problems. European Journal of Operational Research 234(3):658–673.
- Villegas JG, Prins C, Prodhon C, Medaglia AL, Velasco N, 2010 GRASP/VND and multi-start evolutionary local search for the single truck and trailer routing problem with satellite depots. Engineering Applications of Artificial Intelligence 23(5):780–794.
- Villegas JG, Prins C, Prodhon C, Medaglia AL, Velasco N, 2011 A GRASP with evolutionary path relinking for the truck and trailer routing problem. Computers & Operations Research 38(9):1319–1334.
- Villegas JG, Prins C, Prodhon C, Medaglia AL, Velasco N, 2013 A matheuristic for the truck and trailer routing problem. European Journal of Operational Research 230(2):231–244.
- Willemse EJ, Joubert JW, 2016 Constructive heuristics for the mixed capacity arc routing problem under time restrictions with intermediate facilities. Computers & Operations Research 68:30–62.
- Wolsey LA, 1998 Integer Programming (New York: Wiley).
- Xu H, Chen ZL, Rajagopal S, Arunapuram S, 2003 Solving a practical pickup and delivery problem. Transportation Science 37(3):347–364.
- Yang J, Sun H, 2015 Battery swap station location-routing problem with capacitated electric vehicles. Computers & Operations Research 55:217–232.
- Yavuz M, 2017 An iterated beam search algorithm for the green vehicle routing problem. Networks 69(3):317–328.
- Yavuz M, Çapar I, 2017 Alternative-fuel vehicle adoption in service fleets: Impact evaluation through optimization modeling. Transportation Science 51(2):480–493.